

## G OPEN ACCESS

**Citation:** Chen J, Bacelis J, Sole-Navais P, Srivastava A, Juodakis J, Rouse A, et al. (2020) Dissecting maternal and fetal genetic effects underlying the associations between maternal phenotypes, birth outcomes, and adult phenotypes: A mendelian-randomization and haplotype-based genetic score analysis in 10,734 mother–infant pairs. PLoS Med 17(8): e1003305. <u>https://doi.org/</u> 10.1371/journal.pmed.1003305

Academic Editor: Fasil Tekola-Ayele, National Institutes of Health, UNITED STATES

Received: September 12, 2019

Accepted: July 21, 2020

Published: August 25, 2020

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003305

**Copyright:** © 2020 Chen et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Dissecting maternal and fetal genetic effects underlying the associations between maternal phenotypes, birth outcomes, and adult phenotypes: A mendelianrandomization and haplotype-based genetic score analysis in 10,734 mother–infant pairs

Jing Chen<sup>1</sup>, Jonas Bacelis<sup>2,3</sup>, Pol Sole-Navais<sup>2</sup>, Amit Srivastava<sup>4,5</sup>, Julius Juodakis<sup>2</sup>, Amy Rouse<sup>5</sup>, Mikko Hallman<sup>6</sup>, Kari Teramo<sup>7</sup>, Mads Melbye<sup>8,9,10</sup>, Bjarke Feenstra<sup>8</sup>, Rachel M. Freathy<sup>11</sup>, George Davey Smith<sup>12,13,14</sup>, Deborah A. Lawlor<sup>12,13,14</sup>, Jeffrey C. Murray<sup>15</sup>, Scott M. Williams<sup>16</sup>, Bo Jacobsson<sup>2,17</sup>, Louis J. Muglia<sup>4,5\*</sup>, Ge Zhang<sup>4,5\*</sup>

1 Division of Biomedical Informatics, Cincinnati Children's Hospital Medical Center, Department of Pediatrics, University of Cincinnati College of Medicine, Cincinnati, Ohio, United States of America, 2 Department of Obstetrics and Gynecology, Institute of Clinical Sciences, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden, 3 Region Västra Götaland, Sahlgrenska University Hospital, Department of Obstetrics and Gynecology, Gothenburg, Sweden, 4 Division of Human Genetics, Cincinnati Children's Hospital Medical Center, Department of Pediatrics, University of Cincinnati College of Medicine, Cincinnati, Ohio, United States of America, 5 Center for Prevention of Preterm Birth, Perinatal Institute and March of Dimes Prematurity Research Center Ohio Collaborative, Cincinnati Children's Hospital Medical Center, Department of Pediatrics, University of Cincinnati College of Medicine, Cincinnati, Ohio, United States of America, 6 PEDEGO Research Unit and Medical Research Center Oulu, University of Oulu and Department of Children and Adolescents, Oulu University Hospital, Oulu, Finland, 7 Obstetrics and Gynecology, University of Helsinki and Helsinki University Hospital, Helsinki, Finland, 8 Department of Epidemiology Research, Statens Serum Institut, Copenhagen, Denmark, 9 Department of Clinical Medicine, University of Copenhagen, Copenhagen, Denmark, 10 Department of Medicine, Stanford University School of Medicine, Stanford, California, United States of America, 11 Institute of Biomedical and Clinical Science, College of Medicine and Health, University of Exeter, Exeter, United Kingdom, 12 MRC Integrative Epidemiology Unit at the University of Bristol, Bristol, United Kingdom, 13 Population Health Science, Bristol Medical School, University of Bristol, Bristol, United Kingdom, 14 Bristol NIHR Biomedical Research Centre, United Kingdom, 15 Department of Pediatrics, University of Iowa, Iowa City, Iowa, United States of America, 16 Department of Population and Quantitative Health Sciences, Case Western Reserve University School of Medicine, Cleveland, Ohio, United States of America, 17 Department of Genetics and Bioinformatics, Domain of Health Data and Digitalisation, Institute of Public Health, Oslo, Norway

\* ge.zhang@cchmc.org (GZ); louis.muglia@cchmc.org (LJM)

## Abstract

## Background

Many maternal traits are associated with a neonate's gestational duration, birth weight, and birth length. These birth outcomes are subsequently associated with late-onset health conditions. The causal mechanisms and the relative contributions of maternal and fetal genetic effects behind these observed associations are unresolved.

Data Availability Statement: The data used in this study are available to other researchers. In order to respect and protect the interests and privacy of the research participants, the access to the individuallevel phenotype and genotype data requires submitting applications to and approval by the corresponding entities who are in charge of the distribution of the data sets (e.g., FIN, ALSPAC, FIN, MoBa, and dbGaP). This is to ensure that the proposed study aims are consistent with the informed consent under which the data or samples were collected and appropriate data safety and security measures are in place to protect against data breach and unauthorized use. Individual-level phenotype and genotype data from the Finnish (Helsinki) birth cohort are available through the March of Dimes Prematurity Research Center Ohio Collaborative (http://prematurityresearch.org/ ohiocollaborative/), and access will be approved by the Leadership Committee through its director of operations, Joanne Chappell (joanne. chappell@cchmc.org). ALSPAC data are available to scientists on request to the ALSPAC Executive Committee (ALSPAC-exec@bristol.ac.uk) or via website (http://www.bristol.ac.uk/alspac/ researchers/access/), which also provides full details and distributions of the ALSPAC study variables. The detailed policy of data sharing can be found in the ALSPAC data management plan (http://www.bristol.ac.uk/alspac/researchers/dataaccess/documents/alspac-data-management-plan. pdf). MoBa data is available to researchers and research groups at both the Norwegian Institute of Public Health and other research institutions nationally and internationally. The research must adhere to the aims of MoBa and the participants' given consent. All use of data and biological material from MoBa is subject to Norwegian legislation. Terms for applying for access to data and links to the application form and information can be found at https://www.fhi.no/en/studies/ moba/for-forskere-artikler/research-and-dataaccess/. Access to the DNBC (phs000103.v1.p1), HAPO (phs000096.v4.p1), and GPN (phs000714. v1.p1) individual-level phenotype and genetic data can be obtained through dbGaP Authorized Access portal (https://dbgap.ncbi.nlm.nih.gov/dbgap/aa/ wga.cgi?page=login). The informed consent under which the data or samples were collected is the basis for determining the appropriateness of sharing data through unrestricted-access databases or NIH-designated controlled-access data repositories.

**Funding:** This work is supported by a grant from the Burroughs Wellcome Fund (10172896) and a grant from the Cincinnati Children's Hospital Medical Center (GAP/RIP) to GZ and a grant from

## Methods and findings

Based on 10,734 mother-infant duos of European ancestry from the UK, Northern Europe, Australia, and North America, we constructed haplotype genetic scores using single-nucleotide polymorphisms (SNPs) known to be associated with adult height, body mass index (BMI), blood pressure (BP), fasting plasma glucose (FPG), and type 2 diabetes (T2D). Using these scores as genetic instruments, we estimated the maternal and fetal genetic effects underlying the observed associations between maternal phenotypes and pregnancy outcomes. We also used infant-specific birth weight genetic scores as instrument and examined the effects of fetal growth on pregnancy outcomes, maternal BP, and glucose levels during pregnancy. The maternal nontransmitted haplotype score for height was significantly associated with gestational duration ( $p = 2.2 \times 10^{-4}$ ). Both maternal and paternal transmitted height haplotype scores were highly significantly associated with birth weight and length (p  $< 1 \times 10^{-17}$ ). The maternal transmitted BMI scores were associated with birth weight with a significant maternal effect ( $p = 1.6 \times 10^{-4}$ ). Both maternal and paternal transmitted BP scores were negatively associated with birth weight with a significant fetal effect ( $p = 9.4 \times$ 10<sup>-3</sup>), whereas BP alleles were significantly associated with gestational duration and preterm birth through maternal effects ( $p = 3.3 \times 10^{-2}$  and  $p = 4.5 \times 10^{-3}$ , respectively). The nontransmitted haplotype score for FPG was strongly associated with birth weight ( $p = 4.7 \times$ 10<sup>-6</sup>); however, the glucose-increasing alleles in the fetus were associated with reduced birth weight through a fetal effect ( $p = 2.2 \times 10^{-3}$ ). The haplotype scores for T2D were associated with birth weight in a similar way but with a weaker maternal effect ( $p = 6.4 \times 10^{-3}$ ) and a stronger fetal effect ( $p = 1.3 \times 10^{-5}$ ). The paternal transmitted birth weight score was significantly associated with reduced oestational duration ( $p = 1.8 \times 10^{-4}$ ) and increased maternal systolic BP during pregnancy ( $p = 2.2 \times 10^{-2}$ ). The major limitations of the study include missing and heterogenous phenotype data in some data sets and different instrumental strength of genetic scores for different phenotypic traits.

## Conclusions

We found that both maternal height and fetal growth are important factors in shaping the duration of gestation: genetically elevated maternal height is associated with longer gestational duration, whereas alleles that increase fetal growth are associated with shorter gestational duration. Fetal growth is influenced by both maternal and fetal effects and can reciprocally influence maternal phenotypes: taller maternal stature, higher maternal BMI, and higher maternal blood glucose are associated with larger birth size through maternal effects; in the fetus, the height- and metabolic-risk–increasing alleles are associated with increased and decreased birth size, respectively; alleles raising birth weight in the fetus are associated with shorter gestational duration and higher maternal BP. These maternal and fetal genetic effects may explain the observed associations between the studied maternal phenotypes and birth outcomes, as well as the life-course associations between these birth outcomes and adult phenotypes.

the March of Dimes (22-FY17-889) and a grant from the Bill and Melinda Gates Foundation (OPP1175128) to LJM and GZ. The Norwegian Mother and Child Cohort Study (MoBa) is supported by the Norwegian Ministry of Health and Care Services and the Ministry of Education and Research, NIH/NIEHS (contract no N01-ES-75558), and NIH/NINDS (grant no. 1: UO1 NS 047537-01 and grant no. 2: UO1 NS 047537-06A1). The genotyping and analyses were supported by grants from Jane and Dan Olsson Foundation (Gothenburg, Sweden), Swedish Medical Research Council (2015-02559), Norwegian Research Council/FUGE (grant no. 151918/S10; FRI-MEDBIO 249779), March of Dimes (21-FY16-121), and the Burroughs Wellcome Fund Preterm Birth Research Grant (10172896) and by Swedish government grants to researchers in the public health sector (ALFGBG-717501, ALFGBG-507701, ALFGBG-426411) to BJ. RMF is supported by a Sir Henry Dale Fellowship (Wellcome Trust and Royal Society grant: WT104150). BF was supported by a grant from the Oak Foundation. DAL and GDS work in a unit that is supported by the University of Bristol and Medical Research Council (MC\_UU\_00011/1 and MC\_UU\_00011/6). DAL is supported by a grant from the US National Institute of Health (R01 DK10324), an NIHR Senior Investigator Award (NF-0616-10102), a grant from the European Research Council (DevelopObese; 669545) and a grant from the British Heart Foundation (AA/18/7/ 34219). The UK Medical Research Council and Wellcome (grant ref: 102215/2/13/2) and the University of Bristol provide core support for ALSPAC. GWAS data was generated by Sample Logistics and Genotyping Facilities at Wellcome Sanger Institute and LabCorp (Laboratory Corporation of America) using support from 23andMe. The DNBC data sets used for the analyses described in this manuscript were obtained from dbGaP at https://www.ncbi.nlm.nih. gov/gap/ through dbGaP accession number phs000103.v1.p1. The GWAS of Prematurity and its Complications study is one of the genome-wide association studies funded as part of the Gene Environment Association Studies (GENEVA) under the Genes, Environment and Health Initiative (GEI). The HAPO data sets used for the analyses described in this manuscript were obtained from dbGaP at https://www.ncbi.nlm.nih.gov/gap/ through dbGaP accession number phs000096.v4. p1. This study is part of the Gene Environment Association Studies initiative (GENEVA) funded by the trans-NIH Genes. Environment. and Health Initiative (GEI). The GPN datasets used for the analyses described in this manuscript were

## Author summary

### Why was this study done?

- Maternal height, BMI, blood glucose, and blood pressure are associated with gestational duration, birth weight, and birth length. These birth outcomes are subsequently associated with late-onset health conditions.
- The causal mechanisms and the relative contributions of maternal and fetal genetic effects underlying these observed associations are not clear.

### What did the researchers do and find?

- We dissected the relative contributions of maternal and fetal genetic effects using haplotype genetic score analysis in 10,734 mother–infant pairs of European ancestry.
- Genetically elevated maternal height is associated with longer gestational duration and larger birth size. In the fetus, alleles associated with adult height are positively associated with birth size.
- Alleles elevating blood pressure are associated with shorter gestational duration through a maternal effect and are associated with reduced fetal growth through a fetal genetic effect. Alleles that increase blood glucose in the mother are associated with increased birth weight, whereas risk alleles for type 2 diabetes in the fetus are associated with reduced birth weight.
- Alleles raising birth weight in fetus are associated with shorter gestational duration and higher maternal blood pressure during pregnancy.

## What do these findings mean?

- Maternal size and fetal growth are important factors in shaping the duration of gestation.
- Fetal growth is influenced by both maternal and fetal effects. Higher maternal BMI and glucose levels positively associate with birth weight through maternal effects. In the fetus, alleles associated with higher metabolic risks are negatively associated with birth weight.
- More rapid fetal growth is associated with shorter gestational duration and higher maternal blood pressure.
- These maternal and fetal genetic effects can largely explain the observed associations between maternal phenotypes and birth outcomes, as well as the life-course associations between these birth outcomes and adult phenotypes.

obtained from dbGaP at https://www.ncbi.nlm.nih. gov/gap/ through dbGaP accession number phs000714.v1.p1. Samples and associated were provided by the NICHD-funded Genomic and Proteomic Network for Preterm Birth Research (GPN-PBR). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: DAL has received support, via her University, from Roche Diagnostics and Medtronic Ltd for biomarker research unrelated to this study. LJM consults for Mirvie, Inc., a biotech company involved in preterm birth diagnostics. The content of this paper is unrelated to the areas of their diagnostics development. GDS is a member of the Editorial Board of PLOS Medicine. The other authors report no conflicts.

Abbreviations: ALSPAC, The Avon Longitudinal Study of Parents and Children; BMI, body mass index; BP, blood pressure; DBP, diastolic blood pressure; DNBC, The Danish National Birth Cohort; FIN, The Finnish birth data set; FPG, fasting plasma glucose; GPN, The Genomic and Proteomic Network for Preterm Birth Research; GWA, genome-wide association; HAPO, Hyperglycemia and Adverse Pregnancy Outcome; MoBa, The Mother Child data set of Norway; MR, mendelian randomization; MR-PRESSO, MR pleiotropy residual sum and outlier; OGTT, oral glucose tolerance test; PCA, principal components analysis; SBP, systolic blood pressure; SD, standard deviation; SNP, single-nucleotide polymorphism; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology; TSLS, twostage least squares; T2D, type 2 diabetes.

### Introduction

Epidemiological studies have demonstrated that maternal physical and physiological traits associate with birth outcomes. For example, maternal height is positively associated with gestational duration [1, 2], birth weight, and birth length [3, 4]; higher maternal blood glucose is associated with higher birth weight [5]; and elevated maternal blood pressure (BP) is associated with reduced birth weight [6, 7]. These birth outcomes in turn associate with many long-term adverse health outcomes in the offspring, such as obesity [8], type 2 diabetes (T2D) [9], hypertension [10], and cardiovascular diseases [11, 12]. Different mechanisms have been proposed to explain the observed associations between maternal phenotypes and pregnancy outcomes [13-16] (Fig 1A), as well as the life-course associations between birth outcomes and adult phenotypes [17–20] (Fig 1B). Briefly, these include various causal effects (for example, maternal effects, defined as the causal influence of the maternal phenotype on birth outcomes or offspring phenotype [21]), genetically confounded associations due to genetic sharing (between mothers and infants) or shared genetic effects (between a birth outcome and an adult phenotype) [20], and confounding due to the environment. Fetal phenotypes can also affect maternal physiology during or even after pregnancy (fetal drive) [22]. Dissecting these different underlying mechanisms would increase knowledge of the etiology of these critical pregnancy outcomes and provide insights into how pregnancy outcomes are linked with later-onset disorders [13, 23]. Understanding the causal effects of modifiable maternal phenotypes could have implications for clinical interventions to prevent adverse birth outcomes [24]. The shared genetic causes between pregnancy characteristics and late offspring outcomes could provide insights into the molecular pathways through which these shared genetic effects are mediated [20].

Mendelian randomization (MR) [25] studies utilizing maternal genotypes as instrumental variables have been used to probe the causal relationships between maternal phenotypes and pregnancy outcomes [13, 16, 26]. Using this approach, Tyrrell and colleagues [24] demonstrated that higher maternal body mass index (BMI) and blood glucose levels are causally associated with higher birth weight, whereas higher maternal systolic BP (SBP) causes lower birth weight. Using a genome-wide association (GWA) approach, Horikoshi and colleagues [20] demonstrated strong inverse genetic correlations between birth weight and adult cardiometabolic diseases, suggesting a strong genetic component underlying the observed associations between low birth weight and cardiometabolic risks. More recently, Warrington and colleagues estimated maternal and fetal genetic effects on birth weight genome-wide and investigated associations between those genetic effects on birth weight and adult SBP [27].

We previously developed an MR method that utilizes nontransmitted maternal alleles as a valid genetic instrument for maternal phenotypic effects on fetal/offspring outcomes [15]. We showed that the observed association between maternal height and fetal size is mainly due to shared genetics, whereas the association between maternal height and gestational duration is more likely causal. Studies based on this approach have provided novel understandings about the causal relationships between many maternal phenotypes and birth outcomes. They have also highlighted genetic contributions to life-course associations between birth weight and late-onset diseases [20, 27]. However, previous studies have usually examined causal effects of maternal phenotypes on either birth weight or gestational duration separately despite the strong association between them [28, 29]. The studies focusing on birth weight have not explored whether any effects on birth weight are driven by effects on gestational duration because the information was not always available. In addition, the causal effects of fetal growth on gestational duration and maternal phenotypes during pregnancy have not been previously investigated using genetic approaches.

To further our understanding of how various maternal phenotypes are correlated with pregnancy outcomes through maternal or fetal genetic effects and how fetal growth influences



Fig 1. The different mechanisms underlying (A) the associations between maternal phenotypes and pregnancy outcomes and (B) the associations between pregnancy outcomes and late adult phenotypes in offspring. These mechanisms include 1) causal effects of maternal phenotypes on pregnancy outcomes ( $\beta_{XY}^m$ ) and causal effects of pregnancy outcomes on adult phenotypes ( $\beta_{YX'}^r$ ) (green arrows), 2) genetically confounded associations between maternal phenotypes and pregnancy outcomes ( $\beta_{XY}^r$ ) because of genetic sharing between mothers and infants and genetically confounded associations between birth outcomes and adult phenotypes in offspring ( $\beta_{YX'}^r$ ) because of shared genetic effects (blue dashed arrows), 3) confounding due to environmental effects (gray dashed arrows, which were not examined in this study), and 4) fetal drive ( $\beta_{YX}^r$ )—fetus causally influencing maternal phenotypes during pregnancy (red arrow).

https://doi.org/10.1371/journal.pmed.1003305.g001

gestational duration and maternal physiological changes during pregnancy, we expanded our haplotype-based method by considering the mother–fetus duo (pregnancy) as the analytical unit [30] and explicitly modeled maternal and fetal genetic effects using haplotype genetic scores (Fig 2). By testing associations between these haplotype genetic scores and birth outcomes, we systematically investigated the maternal and fetal genetic effects underlying the observed associations between 4 maternal traits (height, BMI, BP, and blood glucose levels) and pregnancy outcomes (gestational duration, birth weight, and birth length). Using this approach, we also examined the associations between fetal growth (using gestational-age–



**Fig 2. Genetic dissection of maternal and fetal genetic effects using haplotype genetic scores in mother-child pairs.** There are 3 groups of alleles (haplotypes) in a mother (M)-fetus (F) duo: the maternal transmitted alleles (h1) can affect a pregnancy outcome through a maternal (1) and/or fetal genetic effect (2), the maternal nontransmitted alleles (h2) can only affect a pregnancy outcome through maternal effect (1), and the paternal transmitted alleles (h3) only through fetal effect (2) (assuming no paternal effect). The paternal transmitted alleles (h3) could influence a maternal phenotype during pregnancy by fetal drive (3). The paternal genetics might be able to influence maternal phenotype and pregnancy outcomes through the environment that the fathers create (i.e., paternal effect or "genetic nurture"). However, for the traits and their associated variants considered in this study (<u>S2 Table</u>), the paternal effects should be minimal and therefore were assumed to be zero.

https://doi.org/10.1371/journal.pmed.1003305.g002

adjusted birth weight as a measure of this) and pregnancy outcomes, maternal BP, and maternal blood glucose levels measured during pregnancy.

#### Methods

A prospective protocol for analysis was not prepared for this study; however, the assembly of the data sets and all the analyses were planned in advance of data analysis. We reported this study according to the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline [31] for cross-sectional studies (S1 STROBE Checklist). All data derived from the present study are presented with or in the paper.

#### Data sets

We used phenotype and genome-wide single-nucleotide polymorphism (SNP) data of 10,734 mother–infant pairs from 6 birth studies (<u>S1 Table</u>). These include 3 case/control data sets

collected from Nordic countries (The Finnish birth data set [FIN], The Mother Child data set of Norway [MoBa], and The Danish National Birth Cohort [DNBC]) for genetic studies of preterm birth [32], a longitudinal birth cohort (the Avon Longitudinal Study of Parents and Children [ALSPAC]) [33] from the UK, a study of preterm birth from the US (The Genomic and Proteomic Network for Preterm Birth Research [GPN]) [34], and the Hyperglycemia and Adverse Pregnancy Outcome [HAPO] study [5] with samples of European ancestry collected from the UK, Canada, and Australia. A detailed description of these data sets can be found in the Supplementary Methods (<u>S1 Text</u>).

We focused on investigating the relationships between maternal height, prepregnancy BMI, BP, blood glucose levels, and pregnancy outcomes including gestational duration (as both quantitative and dichotomous preterm/term trait), birth weight, and birth length. Maternal height, prepregnancy BMI, and the 3 pregnancy outcomes were available in most of the studies (birth weight and length were not available in the MoBa data used here, and birth length was not available in the DNBC data used here) (Table 1). Maternal BP during pregnancy was available in ALSPAC and HAPO. In the HAPO data, BP was measured between 24 and 32 weeks of pregnancy when the mothers underwent an oral glucose tolerance test (OGTT) [5]. In ALSPAC, all BP measurements undertaken during antenatal care were extracted from clinical records; women had a median of 13 (interquartile range 11–16) BP measurements [35]. We used the average of the BPs measured between 30 to 36 weeks of gestation (as close as possible to when the BP was measured in HAPO). Maternal fasting plasma glucose (FPG) levels during pregnancy were available only in the HAPO study. FPG was measured in over 4,000 ALSPAC mothers in a follow-up data collection 18 years after the pregnancy (<u>S1 Text</u> and <u>Table 1</u>).

Because gestational duration is a key determinant of birth weight and birth size, we only included pregnancies with spontaneous deliveries and excluded mother–child pairs without gestational duration information. Pregnancies with known gestational or fetal complications and pre-existing medical conditions were excluded. Detailed inclusion/exclusion criteria are provided in Supplementary Methods (S1 Text). Preterm birth was defined as birth before 37 completed weeks of pregnancy.

This study involves reanalysis of existing data sets, and the proposed analytical aims are consistent with the original consent agreements under which the genomic and phenotypic data were obtained. Therefore, additional ethics approval was not required.

#### Genotype data

Genome-wide SNP data were generated using either Affymetrix 6.0 (Affymetrix, Santa Clara, CA, USA) or various Illumina genotyping arrays (Illumina, San Diego, CA, USA). Standardized genotype quality control procedures were applied to all data sets. Participants of non-European ancestry were identified and excluded using principal components analysis (PCA) (<u>S1 Text</u> and <u>S1 Fig</u>). Genome-wide imputation was performed using Minimac3 [<u>36</u>] and the reference haplotypes from phase 3 of the 1000 Genomes Project [<u>37</u>]. In each data set, the haplotype phasing was done in all maternal and fetal samples using Shapeit2 [<u>38</u>]. This program accommodates mother-child relationship and accurately estimates mother-child allele transmission when phasing mother-child duos together.

#### **Construction of genetic scores**

We constructed weighted genetic scores to instrument various maternal phenotypes using GWA SNPs and their estimated effect sizes reported by the most recent large GWA studies (<u>S1</u> <u>Text</u> and <u>S2 Table</u>). Specifically, 2,130 height-associated SNPs and 628 BMI-associated SNPs reported by the GIANT consortium [<u>39</u>] were used to build genetic scores for height and BMI,

Trait		FIN			MoBa			DNBC			HAPO			GPN			ALSPAC			Total	
	Z	Mean	SD	z	Mean	SD	Z	Mean	SD	z	Mean	SD	z	Mean	SD	Z	Mean	SD	z	Mean	SD
Maternal traits																					
Height (cm)	1,170	166.7	5.8	975	167.9	5.8	1,653	168.9	6.1	1,089	164.5	6.2	342	164.9	7.8	4,993	164.4	6.6	10,222	165.7	6.4
BMI (kg/m <sup>2</sup> )	1,168	23.0	3.7	958	24.0	4.2	1,625	23.5	4.2	1,051	24.2	4.7	337	24.2	5.3	4,759	22.9	3.7	9,898	23.3	4.0
SBP (mmHg)		NA			NA			NA		1,089	108.0	9.7		NA		4,923	113.1	6.8	6,012	112.1	7.4
DBP (mmHg)										1,089	71.1	7.8				4,943	66.8	4.7	6,032	67.6	5.4
FPG (mmol/L) <sup>a</sup>										1,088	4.5	0.4				2,452	5.2	0.4	3,540	5.0	0.4
Birth outcomes <sup>b</sup>																					
Gestational days	892	282.0	6.3	522	280.4	3.6	977	282.6	3.0	1,034	281.2	7.2	197	276.4	4.2	4,967	279.8	7.9	8,589	280.5	7.0
	255	237.9	13.4	487	248.2	11.4	748	240.1	15.2		None		146	212.0	20.3	212	242.7	13.5	1,848	240.0	14.3
Birth weight (g)	892	3,578.0	434.3		NA		973	3,712.6	456.5	1,034	3,463.3	487.0	197	3,505.2	366.0	4,909	3,514.3	455.7	8,005	3,538.7	455.5
	255	2,348.8	477.1				739	2,459.5	645.8		None		146	1,641.4	522.1	208	2,543.1	536.8	1,348	2,362.9	587.4
Birth length (cm)	892	50.4	1.9		NA			NA		1,034	50.7	2.2	192	34.7	1.4	4,004	51.0	2.1	6,122	50.4	2.1
	253	45.0	2.9								None		146	28.4	3.5	152	47.3	2.5	551	41.3	3.0
Male/Female <sup>c</sup>		606/564			503/506			908/831			550/539			189/154		2	,690/2,694	4	5,	446/5,288	
<sup>a</sup> In HAPO, matern	al FPG v	vas measu	red betw	veen 24	–32 weel	ks of ge.	station. l	n ALSPA	C, the F	PG was 1	measured	18 years	after pı	regnancy.							
<sup>b</sup> Descriptive statist	ics of pr	sgnancy o	utcome	s were c	alculatec	l in teri	n (≥38 1	veeks, up]	ber row)	) and pre	term infan	1ts (<37	weeks,	lower rov	v) separ	ately. Th	iere were i	no pretei	rm infant	s in the H	APO
data set.																					
<sup>c</sup> Number of male a	nd fema	le infants.																			
Abbreviations: AL	SPAC, 7	The Avon	Longitu	dinal St	udy of P	arents ¿	und Chil	dren; BM	l, body 1	nass inde	ex; DBP, d	liastolic t	d poole	ressure; I	DNBC, 1	The Dani	ish Nation	al Birth	Cohort; F	IN, The F	innish

Table 1. Descriptive statistics of maternal traits and birth outcomes.

birth data set; FPG, fasting plasma glucose; GPN, The Genomic and Proteomic Network for Preterm Birth Research; HAPO, The Hyperglycemia and Adverse Pregnancy Outcome Study; MoBa, The Mother Child data set of Norway; NA, not available; SBP, systolic blood pressure; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1003305.t001

respectively. Eight hundred thirty-one SNPs associated with BP [40] were used to build genetic scores for BP. For simplicity, we built a score for average BP using the mean estimated effects of SBP and diastolic BP (DBP). For FPG, we used 22 SNPs associated with FPG levels identified in individuals without diabetes [41]. We also constructed a T2D genetic score using 306 T2D SNPs [42] (excluding SNPs overlapping or in close linkage disequilibrium with the 22 FPG SNPs) (S1 Text and S2 Table). To examine the associations of fetal growth (as proxied by birth weight) with pregnancy outcomes and maternal BP and FPG during pregnancy, we constructed genetic scores using 86 SNPs associated with birth weight with confirmed fetal effect [27]. The lists of these GWA SNPs used in constructing genetic scores are provided in S1 Data.

For each set of GWA SNPs, we constructed 2 genotype genetic scores:  $S_{mat}$  (maternal genotype score),  $S_{fet}$  (fetal genotype score), and 3 haplotype genetic scores:  $S_{h1}$ ,  $S_{h2}$ , and  $S_{h3}$ , respectively, based on the maternal transmitted (h1), maternal nontransmitted (h2), and paternal transmitted alleles (h3) (Fig\_2). It follows that  $S_{mat} = S_{h1} + S_{h2}$  and  $S_{fet} = S_{h1} + S_{h3}$ .

#### Statistical analyses

**Phenotype associations and instrumental strength of genetic scores.** We first assessed the associations between the 4 maternal phenotypes (X) (i.e., height, BMI, BP, and FPG) and each pregnancy outcome (Y) (i.e., gestational duration, preterm birth, birth weight, and birth length) using regression analyses. Maternal age, fetal sex, maternal height, and prepregnancy BMI were included as covariates. Because gestational duration influences birth weight and length in a nonlinear fashion, the first 3 orthogonal polynomials of gestational duration were included as covariates in the analysis of birth weight and length. These analytical models are described in more detail in the Supplementary Methods (<u>S1 Text</u>).

The instrumental strength of the genetic scores was checked by the variance in a maternal phenotype explained ( $R^2$ ) by the corresponding genetic scores.

Association tests between haplotype genetic scores and pregnancy outcomes. Associations between the haplotype genetic scores and the pregnancy outcomes were tested using regression models like those used in the association analysis described above, except that the maternal phenotypes (X) were replaced by their corresponding 3 haplotype genetic scores ( $S_{h1} + S_{h2} + S_{h3}$ ). The associations between these haplotype scores can differentiate between maternal and fetal genetic effects (Fig 2). Specifically, an association of  $S_{h2}$  (maternal nontransmitted haplotype score) with a pregnancy outcome suggests a maternal (intrauterine phenotypic) effect, whereas an association of  $S_{h3}$  (paternal transmitted haplotype score) with the pregnancy outcomes suggests fetal genetic effects. The 3 haplotype genetic scores of the same mother-child pairs ( $S_{h1}$ ,  $S_{h2}$ , and  $S_{h3}$ ) were simultaneously tested in a single-regression model (i.e., Y is modeled as a function of  $S_{h1} + S_{h2} + S_{h3} + Cov$ , where Cov is a list of appropriate covariates), and hence, they had exactly the same sample size. Therefore, the effect size estimates of these haplotype scores and their associated *p*-values can be directly compared to assess the directions and relative contributions of the maternal and fetal effects.

**Modeling of maternal and fetal genetic effects.** Whereas  $S_{h2}$  and  $S_{h3}$  can be used to draw inference about maternal and fetal genetic effects, this question can be addressed with greater statistical power by also including  $S_{h1}$ , the maternal transmitted haplotype score, in the model. Thus, we modeled the maternal effect and fetal genetic effect as different linear combinations of the regression coefficients of the 3 haplotype genetic scores (Fig 2 and S1 Text) [30]. Under the assumptions of additivity between maternal and fetal effect and zero parent-of-origin effect, the total effect ( $\beta_{h1}$ ) of the maternal transmitted haplotype (h1) should be equal to the summation of the maternal effect ( $\beta_{h2}$ ) of the nontransmitted haplotype (h2) and fetal genetic effect ( $\beta_{h3}$ ) of the paternal transmitted haplotype (h3). Thus, ( $\beta_{h1}-\beta_{h3}$ ) and ( $\beta_{h1}-\beta_{h2}$ ),

respectively, represent the maternal effect and the fetal genetic effect of the maternal transmitted haplotype (h1). Therefore, the average maternal effect ( $\beta_{MY}$ ) of the 2 maternal haplotypes on a birth outcome (Y) can be expressed as ( $\beta_{h1}-\beta_{h3}+\beta_{h2}$ )/2, and the average fetal genetic effect ( $\beta_{FY}$ ) of the maternal and paternal transmitted haplotypes can be expressed as ( $\beta_{h1}-\beta_{h2}+\beta_{h3}$ )/2. Because these linear combinations also capture the maternal or fetal genetic effects of the maternal transmitted haplotype (h1), they are more powerful than the methods only using the maternal nontransmitted haplotype (h2) or the paternal transmitted haplotype (h3) as instruments, respectively, for maternal effect and fetal genetic effect.

Estimation of maternal causal effects. The estimated maternal effect ( $\beta_{MY}$ ) from the haplotype genetic score association analyses can be interpreted as the amount of change in a pregnancy outcome (Y) caused by a certain amount of difference in a maternal phenotype (X) associated with one-unit genetic score. The maternal causal effect ( $\hat{\beta}_{XY}^m$ ) was estimated using the ratio estimate [43] ( $\hat{\beta}_{MY}/\hat{\beta}_{MX}$ ), where  $\hat{\beta}_{MX}$  is the estimated maternal effect on the maternal phenotype (Fig 2). As an alternative, we also performed instrumental variable analysis using the two-stage least-squares (TSLS) approach [43], with the maternal nontransmitted haplotype score ( $S_{h2}$ ) as the genetic instrument for maternal causal effect [15].

**Estimation of genetically confounded associations.** The fetal genetic effect  $(\beta_{FY})$  reflects that the genetic variants associated with an adult phenotype (X') in the offspring or the corresponding maternal phenotype (X) have direct fetal genetic effect on a pregnancy outcome (Y). This shared genetic effect can confound the association between a maternal phenotype (X) and a pregnancy outcome (Y), as well as the association between the pregnancy outcome (Y) and the adult phenotype (X') in offspring (Fig 1).

By assuming that all the genetic variants associated with an adult phenotype in offspring (X') or the corresponding maternal phenotype (X) have a similar effect on a pregnancy outcome (Y) as the fetal genetic effect estimated from the genetic score built on known GWA SNPs ( $\hat{\beta}_{FY}$ ), we can approximately estimate the magnitude of these genetically confounded associations (see <u>S1 Text</u> for details). Specifically, the genetically confounded association between a maternal phenotype (X) and a pregnancy outcome (Y) due to the shared genetic effect can be estimated by

$$\frac{\hat{\beta}_{XY}^c = \frac{h_X^2}{2}\hat{\beta}_{FY}}{\hat{\beta}_{MX}},$$

where  $\hat{\beta}_{MX}$  is estimated maternal effect and  $h_X^2$  is the heritability (the proportion of additive genetic variance) of the maternal phenotype (X).

Similarly, the genetically confounded association between a pregnancy outcome (Y) and an adult (late) phenotype (X') in offspring can be estimated by

$$\hat{\beta}_{YX'}^{\epsilon} = h^2 \frac{\operatorname{Var}(X')}{\operatorname{Var}(Y)} \left( \frac{\hat{\beta}_{MY}}{2 + \hat{\beta}_{FY}} \right)$$
(Method 1)

and

$$\hat{\beta}_{YX'}^{c} = h^2 \frac{\operatorname{Var}(X')}{\operatorname{Var}(Y)} \left( \frac{\widehat{\beta_{h1}} + \widehat{\beta_{h3}}}{2} \right),$$
(Method 2)

where  $h^2$  is the heritability of the adult phenotype and Var(X') and Var(Y) are, respectively, the variance of the adult phenotype and the variance of the pregnancy outcome. The first method

(Method 1) can partition the confounded association into the maternal  $\left(\frac{\beta_{MY}}{2}\right)$  and the fetal component ( $\beta_{FY}$ ) (S1 Text).

**Multivariable MR analysis.** The genetic scores built on hundreds or thousands of SNPs are likely to be less specific because they are more likely to be associated with other phenotypic traits [44], which can introduce ambiguities in the interpretation of genetic score associations [45]. To circumvent this issue, we performed a two-sample multivariable MR analysis [46, 47] using the MR pleiotropy residual sum and outlier (MR-PRESSO) [48] to detect and correct for variants with horizontal pleiotropic effects [49] in multiple-variant MR testing. Three MR-PRESSO tests were applied: the global test was used to detect the presence of horizontal pleiotropy, the outlier test to identify variants with significant horizontal pleiotropic effects, and the distortion test to estimate the distortion caused by significant horizontal pleiotropic outlier variants. This analysis studies the maternal or fetal genetic effects by testing whether the effects of the maternal transmitted (h1), maternal nontransmitted (h2), and paternal transmitted (h3) alleles of the GWA SNPs on a pregnancy outcome are proportional to their reported effects on an adult phenotype (X') in the reference GWA studies (S1 Text and S4 Fig). The allele-specific effect estimates (for the h1, h2, and h3 alleles) of each SNP on a pregnancy outcome were obtained using the same regression methods for haplotype genetic score analysis (S1 Text).

We did meta-analyses of the results from all available data sets to generate the overall results. Fixed-effect meta-analysis was used to combine the regression coefficients and standard errors from individual studies, and we checked between-study heterogeneity using Cochran's Q test. The meta-analyses were done using the R metafor package [50].

#### Results

## Phenotypic associations between maternal phenotypes and pregnancy outcomes

We used 10,734 mother–infant pairs with both genotype and phenotype data in our analyses (<u>S1 Table</u>). Distributions of key variables for the maternal phenotypes and pregnancy outcomes are shown in <u>Table 1</u> and <u>S5–S7</u> Figs.

The meta-analysis across the 6 data sets showed that taller maternal height was associated with longer gestational duration (0.14 day/cm, 95% CI: 0.10 to 0.18,  $p = 2.2 \times 10^{-12}$ ), lower preterm birth risk (OR = 0.97 /cm, 95% CI: 0.96 to 0.98,  $p = 2.2 \times 10^{-9}$ ), and higher birth weight (15 g/cm, 95% CI: 13.7 to 16.3,  $p = 1.5 \times 10^{-111}$ ) and length (0.068 cm/cm, 95% CI: 0.061 to 0.075,  $p = 1.6 \times 10^{-75}$ ). Maternal BMI was positively associated with birth weight (15.6 g/[kg/m<sup>2</sup>], 95% CI: 13.5 to 17.7,  $p = 1.0 \times 10^{-47}$ ) and birth length (0.05 cm/[kg/m<sup>2</sup>], 95% CI: 0.04 to 0.06,  $p = 3.9 \times 10^{-15}$ ) but was not associated with gestational duration (0.05 day per kg/m<sup>2</sup>, 95% CI: -0.01 to 0.11, p = 0.12) or preterm birth risk (OR = 0.99 per kg/m<sup>2</sup>, 95% CI: 0.98 to 1.01, p = 0.42) (S3 Table).

Using data from ALSPAC and HAPO, we observed that maternal BP during pregnancy was negatively associated with gestational duration and birth weight. The estimated effect sizes on gestational duration by SBP and DBP were -0.04 day/mmHg (95% CI: -0.08 to -0.01,  $p = 7.3 \times 10^{-3}$ ) and -0.11 day/mmHg (95% CI: -0.15 to -0.06,  $p = 3.3 \times 10^{-6}$ ), respectively. The estimated effect sizes on birth weight by SBP and DBP were -3.0 g/mmHg (95% CI: -4.6 to -1.5,  $p = 1.8 \times 10^{-4}$ ) and -6.2 g/mmHg (95% CI: -8.4 to -3.9,  $p = 6.0 \times 10^{-8}$ ), respectively. In HAPO, there was a strong positive association between maternal FPG and birth weight (192 g/[mmol/L], 95% CI: 116 to 268,  $p = 5.7 \times 10^{-7}$ ) and birth length (0.62 cm/[mmol/L], 95% CI: 0.27 to 0.97,  $p = 4.8 \times 10^{-4}$ ). However, the association between FPG measured 18 years after pregnancy with either birth weight or length in the ALSPAC data set was close to zero with wide confidence intervals (S3 Table).

#### Associations between genetic scores and maternal phenotypes

We examined the instrumental strength of the genetic scores for the various maternal phenotypes. The maternal genotype genetic scores ( $S_{mat}$ ) were associated with the corresponding maternal phenotypes and explained a substantial fraction of the phenotypic variances with similar contributions from the transmitted (h1) or the nontransmitted haplotype scores (h2) (S4 Table).

The maternal height genotype score ( $S_{mat}$ ) explained >20% of the maternal height variance (<u>S5 Table</u>), and the maternal BMI genotype score ( $S_{mat}$ ) explained approximately 5% of the maternal BMI variance (<u>S6 Table</u>). The BP genotype scores explained over 2% variance in maternal BP (<u>S7 Table</u>), which is less than half of the reported fraction of variance explained by the same score (5.7%) in a published GWA study of nonpregnant women and men (generally of an older age than pregnant women) [<u>40</u>], suggesting that these BP SNPs have a larger effect on BP in older populations or a weaker effect on maternal BP during pregnancy.

In HAPO, the maternal FPG genetic score built on 22 SNPs explained 8.3% of the FPG variance measured between 24–32 weeks. In ALSPAC, the same score explained 4.1% of the variance of FPG measured 18 years after pregnancy. By contrast, the T2D score (306 SNPs) explained much less FPG variance (<u>S8 Table</u>).

For each phenotype, we checked the correlations among the various genotype and haplotype genetic scores (<u>S9 Table</u>). For height, we observed significant correlations between the maternal genotype ( $S_{mat}$ ) and the paternal transmitted haplotype score ( $S_{h3}$ ) and between the maternal transmitted ( $S_{h1}$ ) and nontransmitted scores ( $S_{h2}$ ), indicating assortative mating [<u>15</u>, <u>51</u>, <u>52</u>] and increased homozygosity of height-associated SNPs.

## Maternal causal effects and genetically confounded associations between maternal phenotypes and birth outcomes

We next utilized haplotype genetic scores as genetic instruments to dissect the maternal and fetal genetic effects underlying the observed associations between maternal phenotypes and pregnancy outcomes (Table 2). Detailed meta-analysis of individual data sets can be found in <u>S8–S12</u> Figs. We also conducted random-effects meta-analyses (<u>S10 Table</u>), and the results were essentially the same as the results obtained by fixed-effect meta-analyses (<u>Table 2</u>). To further check the robustness of the results, we conducted the analyses separately in ALSPAC (<u>S11 Table</u>) and the other 5 data sets (<u>S12 Table</u>). The results were similar, except the 5 data sets showed more associations with preterm birth, probably because these data sets had more preterm pregnancies. Based on the estimated maternal and fetal genetic effects, we estimated the maternal causal effects and genetically confounded associations between maternal phenotypes and birth outcomes due to shared genetics (Methods and Fig 3).

**Maternal height.** The maternal nontransmitted height genetic score  $(S_{h2})$  was positively associated with gestational duration  $(p = 2.2 \times 10^{-4})$  and negatively associated with preterm birth  $(p = 9.7 \times 10^{-4})$  (Table 2). The ratio estimates showed a maternal causal effect of approximately 1.0 days (95% CI: 0.38 to 1.64,  $p = 1.8 \times 10^{-3}$ ) longer gestation per 1-standard deviation (SD) (6.4 cm) increase in maternal height. This effect was offset by a weaker and opposite fetal genetic effect of 0.71 days' (95% CI: 0.07 to 1.35,  $p = 2.9 \times 10^{-2}$ ) shorter gestation per the same amount of genetic score associated with a 1-SD increase in maternal height (Fig 3 and S13 Table).

Maternal and paternal transmitted haplotype scores ( $S_{h1}$  and  $S_{h3}$ ) for height were positively associated with birth weight and birth length. The maternal nontransmitted score ( $S_{h2}$ ) was also positively associated with birth weight and length, but the effect estimates were smaller

Maternal Trait (Unit) Haplotype Score		tationa	l Days	Preter	m Birth	(log[OR])	Bi	rth V	Veight (g)	Bi	rth Leng	th (cm)
Tests <sup>a</sup>	Beta <sup>c</sup>	SE	p-Value	Beta	SE	p-Value	Beta	SE	p-Value	Beta	SE	p-Value
Height (cm)												
<i>Maternal transmitted</i> ( $\beta_{h_1}$ )	0.038	0.046	0.41	-0.027	0.011	0.016*	20	1.6	$6.10\times10^{-38_{\ast}}$	0.097	0.0087	$1.10  imes 10^{-28_{st}}$
Maternal nontransmitted ( $\beta_{h2}$ )	0.17	0.047	0.00022*	-0.037	0.011	0.00097*	6.3	1.6	$7.60\times10^{-5*}$	0.022	0.0088	0.011*
Paternal transmitted ( $\beta_{h3}$ )	-0.041	0.046	0.37	0.023	0.011	0.038*	14	1.6	$3.10\times10^{-19*}$	0.074	0.0085	$4.70\times10^{-18*}$
Maternal effect ( $\beta_{MY}$ )	0.13	0.04	0.0017*	-0.044	0.0098	$9.00 \times 10^{-6*}$	6.3	1.4	$5.20\times 10^{-6_{\ast}}$	0.022	0.0076	0.0033*
<i>Fetal effect (</i> $\beta_{FY}$ <i>)</i>	-0.089	0.041	0.029*	0.016	0.0099	0.1	14	1.4	$1.10\times10^{-23_{\ast}}$	0.074	0.0076	$2.70\times10^{-22*}$
BMI (kg/m <sup>2</sup> )												
<i>Maternal transmitted</i> ( $\beta_{h1}$ )	-0.059	0.17	0.73	-0.065	0.041	0.11	23	5.8	$7.00 \times 10^{-5*}$	0.066	0.032	0.038*
Maternal nontransmitted ( $\beta_{h2}$ )	-0.11	0.17	0.52	0.015	0.041	0.71	8.8	5.8	0.13	0.079	0.032	0.013*
Paternal transmitted ( $\beta_{h3}$ )	0.027	0.17	0.88	0.049	0.04	0.23	-6.5	5.7	0.26	0.027	0.032	0.39
Maternal effect ( $\beta_{MY}$ )	-0.098	0.15	0.51	-0.048	0.036	0.18	19	5	0.00016*	0.06	0.028	0.031*
<i>Fetal effect</i> ( $\beta_{FY}$ )	0.043	0.15	0.77	-0.018	0.036	0.6	4	5	0.43	0.0077	0.028	0.78
BP <sup>b</sup> (mmHg)												
<i>Maternal transmitted</i> ( $\beta_{h1}$ )	-0.22	0.064	0.00067*	0.034	0.015	0.027*	-6.8	2.2	0.0016*	-0.018	0.012	0.13
Maternal nontransmitted ( $\beta_{h2}$ )	-0.035	0.064	0.59	0.047	0.016	0.0023*	-3.1	2.2	0.16	-0.018	0.012	0.13
Paternal transmitted ( $\beta_{h3}$ )	-0.016	0.064	0.8	0.005	0.015	0.74	-5.9	2.1	0.0053*	-0.017	0.012	0.15
<i>Maternal effect (<math>\beta_{MY}</math>)</i>	-0.12	0.055	0.033*	0.038	0.013	0.0045*	-2	1.9	0.27	-0.0096	0.01	0.35
<i>Fetal effect</i> ( $\beta_{FY}$ )	-0.1	0.056	0.075	-0.0035	0.013	0.8	-4.8	1.9	0.0094*	-0.0086	0.01	0.4
FPG (mmol/L)												
<i>Maternal transmitted</i> ( $\beta_{h1}$ )	-3.9	1.8	0.029*	0.6	0.43	0.16	13	59	0.82	-0.067	0.32	0.84
Maternal nontransmitted ( $\beta_{h2}$ )	-3.2	1.8	0.071	0.54	0.43	0.21	270	59	$4.70\times10^{-6*}$	0.71	0.32	0.029*
Paternal transmitted ( $\beta_{h3}$ )	2.7	1.7	0.12	-0.38	0.43	0.37	-52	59	0.38	-0.13	0.32	0.69
<i>Maternal effect (<math>\beta_{MY}</math>)</i>	-5	1.5	0.0012*	0.77	0.37	0.039*	170	51	0.0011*	0.37	0.28	0.19
<i>Fetal effect (</i> $\beta_{FY}$ <i>)</i>	0.99	1.5	0.51	-0.14	0.36	0.7	-150	50	0.0022*	-0.46	0.28	0.096
T2D (log[OR])												
<i>Maternal transmitted</i> ( $\beta_{h1}$ )	0.013	0.3	0.97	-0.0079	0.074	0.91	-14	10	0.17	-0.065	0.056	0.25
Maternal nontransmitted ( $\beta_{h2}$ )	0.05	0.31	0.87	0.013	0.072	0.86	33	10	0.0012*	0.059	0.057	0.3
Paternal transmitted ( $\beta_{h3}$ )	0.83	0.31	0.0069*	-0.11	0.073	0.13	-28	10	0.0061*	-0.033	0.057	0.56
Maternal effect ( $\beta_{MY}$ )	-0.39	0.27	0.15	0.058	0.063	0.36	24	8.9	0.0064*	0.015	0.049	0.76
Fetal effect ( $\beta_{FY}$ )	0.4	0.27	0.14	-0.067	0.062	0.28	-39	8.8	$1.30 \times 10^{-5*}$	-0.085	0.049	0.082

#### Table 2. Association between haplotype genetic scores and birth outcomes.

<sup>a</sup> $\beta_{h_1}$ ,  $\beta_{h_2}$ , and  $\beta_{h_3}$  are the effects of the 3 haplotype genetic scores.  $\beta_{MY} = (\beta_{h_1} - \beta_{h_3} + \beta_{h_2})/2$  and  $\beta_{FY} = (\beta_{h_1} - \beta_{h_2} + \beta_{h_3})/2$  are, respectively, the maternal and fetal genetic effects modeled by linear combinations of the haplotype effects.

<sup>b</sup>BP: mean of the SBP and DBP scores.

<sup>c</sup>Beta: estimated effects of genetic score associations given by per unit change in genetic scores of the maternal traits.

\**p*-Values less than 0.05.

Abbreviations: BMI, body mass index; BP, blood pressure; DBP, diastolic BP; SBP, systolic BP; SE, standard error; T2D, type 2 diabetes.

https://doi.org/10.1371/journal.pmed.1003305.t002

than the transmitted haplotype scores (Table 2). The larger effects of transmitted alleles indicate height-associated SNPs can influence growth in early prenatal development through fetal genetic effect. The estimates for maternal causal and fetal genetic effects were 50 g (95% CI: 29 to 72,  $p = 5.6 \times 10^{-6}$ ) and 111 g (95% CI: 89 to 133,  $p = 5.0 \times 10^{-23}$ ), respectively, for birth weight and 0.18 cm (95% CI: 0.06 to 0.30,  $p = 3.3 \times 10^{-3}$ ) and 0.59 cm (95% CI: 0.47 to 0.71,  $p = 1.0 \times 10^{-21}$ ) for birth length per genetic alleles associated with a 1-SD (6.4 cm) increase in maternal height (Fig 3, S13 Fig, and S13 Table). **Maternal prepregnancy BMI.** BMI haplotype genetic scores were not significantly associated with gestational duration or preterm birth (Table 2), and the ratio estimates of maternal causal effects on gestational duration (-0.45 day, 95% CI: -1.80 to 0.89, p = 0.51) and preterm birth (OR = 0.8, 95% CI: 0.6 to 1.1, p = 0.18) were not significant (S13 Table), suggesting both minimal maternal and fetal effect of the BMI-associated SNPs on gestational duration. Linear hypotheses modeling suggested that BMI-associated SNPs have some maternal and no fetal effect on birth weight and length (Table 2). The estimated maternal causal effect on birth weight was 88 g (95% CI: 42 to 134,  $p = 1.9 \times 10^{-4}$ ) and on birth length was 0.28 cm (95% CI: 0.02 to 0.53,  $p = 3.2 \times 10^{-2}$ ) per 1-SD (4.0 kg/m<sup>2</sup>) genetically increased BMI (Fig 3, S13 Fig, and S13 Table).

**Maternal BP.** There was a significant association between the maternal nontransmitted BP genetic score and increased preterm birth risk ( $p = 2.3 \times 10^{-3}$ ), suggesting the association between maternal BP and gestational duration was primarily driven by a maternal effect (Table 2). The estimated causal effect sizes based on the ratio method were –2.3 days (95% CI: -4.4 to –0.14,  $p = 3.6 \times 10^{-2}$ ) on gestational duration and OR = 2.1 (95% CI: 1.2 to 3.5,  $p = 5.7 \times 10^{-3}$ ) in preterm birth risk per 1-SD (5.8 mmHg) genetically increased maternal BP (Fig 3, S13 Fig, and S13 Table).

Both maternal transmitted and paternal transmitted BP scores were negatively associated with birth weight ( $p = 1.6 \times 10^{-3}$  and  $5.3 \times 10^{-3}$  for  $S_{h1}$  and  $S_{h3}$ , respectively; <u>Table 2</u>), suggesting the negative association between maternal BP and birth weight was mainly caused by a fetal genetic effect. The estimated fetal genetic effect was a reduction of 94 g (95% CI: 21 to 166,  $p = 1.1 \times 10^{-2}$ ) in birth weight by alleles associated with a 1-SD (5.8 mmHg) increase in maternal BP (Fig 3 and <u>S13 Table</u>).

**Maternal FPG.** We observed a positive association between maternal nontransmitted FPG genetic score ( $S_{h2}$ ) and birth weight ( $p = 4.7 \times 10^{-6}$ ), indicating a strong causal effect of increased maternal FPG level on higher birth weight (Table 2). In the HAPO data set, the estimated causal effect size by TSLS based on ( $S_{h2}$ ) was 147 g (95% CI: 15 to 279,  $p = 3.0 \times 10^{-2}$ ) per 1-SD (0.36 mmol/L) increase in maternal FPG, and the ratio estimate in all mother–child pairs was 59 g (95% CI: 22 to 96,  $p = 1.8 \times 10^{-3}$ ) (S13 Table). Interestingly, but not unexpectedly, the linear modeling showed a negative fetal genetic effect of FPG increasing alleles on birth weight (-54 g, 95% CI: -90 to -18,  $p = 3.3 \times 10^{-3}$ ). The linear hypothesis modeling also showed a significant negative maternal effect of high maternal FPG on gestational duration (-1.7 days, 95% CI: -2.8 to -0.6,  $p = 2.0 \times 10^{-3}$ ) and increased risk for preterm birth (OR = 1.3, 95% CI: 1.0 to 1.7,  $p = 4.3 \times 10^{-2}$ ) per 1-SD (0.36 mmol/L) increase in maternal FPG (Fig 3, S13 Fig, and S13 Table).

To further understand whether the observed associations were driven by the SNPs that influence normal FPG levels or by the SNPs associated with pathological hyperglycemia (i.e., T2D), we tested associations between the T2D haplotype scores and pregnancy outcomes. Compared to the FPG scores, the maternal nontransmitted T2D score ( $S_{h2}$ ) was less significantly associated with both birth weight and gestational duration, but the negative associations between the paternal transmitted score ( $S_{h3}$ ) and birth weight and gestational duration were more apparent (Table 2).

As shown in Fig 3 and S13 Fig, the combinations of the estimated maternal causal effects and the genetically confounded associations due to genetic transmission were consistent with the observed phenotypic associations between maternal phenotypes and birth outcomes. The maternal causal effects were usually more dominant than the genetically confounded associations in shaping the phenotypic associations between maternal phenotypes and birth outcomes. In some cases, the maternal causal effects and the genetically confounded associations pointed to opposite directions (for example, between height and gestational duration and



Fig 3. Observed phenotypic associations, estimated maternal causal effects, fetal genetic effects, and genetically confounded associations per 1-SD change in maternal traits on gestational duration (left) and birth weight (adjusted by gestational duration) (right). The 1-SD values for maternal traits are 6.4 cm (height), 4.0 kg/m<sup>2</sup> (BMI), 5.8 mmHg (BP), and 0.36 mmol/L (FPG). \* indicates birth weight adjusted by gestational duration. BMI, body mass index; BP, blood pressure; FPG, fasting plasma glucose; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1003305.g003

between FPG and birth weight), and in these situations, the estimated maternal causal effects could be larger than the observed phenotypic associations.

# Genetically confounded associations between birth outcomes and adult phenotypes in offspring

We estimate the magnitude of genetically confounded associations between the birth outcomes and adult phenotypes (in the offspring) based on the hypothesis that the association was driven by the variants that were associated with an adult phenotype were also associated with a birth outcome (Methods). The 2 methods (Method 1 and Method 2) generated similar results (S14 Table). As shown in Fig 4 and S14 Table, a 1-SD change in birth weight (gestational-ageadjusted) was estimated to be associated with 0.20 SD (95% CI: 0.17 to 0.24,  $p = 3.4 \times 10^{-28}$ ) and 0.076 SD (95% CI: 0.014 to 0.138,  $p = 1.6 \times 10^{-2}$ ) differences, respectively, in adult height and BMI. One-SD increases in both gestational duration and birth weight were estimated to be associated with a 0.05 SD decrease in adult BP and a 0.025–0.03 SD decrease in adult FPG level, and birth weight was also estimated to be negatively associated with susceptibility to T2D (OR = 0.91, 95% CI: 0.85 to 0.98,  $p = 7.3 \times 10^{-3}$ ). The genetically confounded associations between birth outcomes and adult phenotypes (S14 Fig and S15 Fig) were mainly driven by the shared genetic effects in offspring; however, the maternal effects could substantially confound the associations and point to opposite directions to the confounded associations by fetal



Estimated differences in addit phenotypes (SD)

Fig 4. Estimated differences in adult phenotypes (in SDs) per 1-SD difference in birth weight and gestational duration. The birth weight was adjusted by gestational duration; 1 SD = 426 g. 1 SD of gestational duration was 11.4 days. The 1-SD values for adult phenotypes were assumed to be 6.4 cm (height), 4.0 kg/m<sup>2</sup> (BMI), 6.9 mmHg (BP), 0.37 mmol/L (FPG), and 1.81 for log OR of T2D. \* indicates birth weight adjusted by gestational duration. BMI, body mass index; BP, blood pressure; FPG, fasting plasma glucose; SD, standard deviation; T2D, type 2 diabetes.

https://doi.org/10.1371/journal.pmed.1003305.g004

genetic effects in certain cases (for example, the confounded associations between gestational duration and height and between birth weight and FPG or T2D risk). We compared the genetically confounded associations between birth weight and adult phenotypes with the estimated phenotypic associations from genetic correlation [20], the observed associations in the ALSPAC data, and the reported associations from an epidemiological meta-analysis [53] (Fig 5). The estimated genetically confounded associations using our approach were similar to those estimated from genetic correlations [20] and were largely consistent with the observed or reported associations between birth weight and adult phenotypes with the exception of body height, for which the observed association (from the ALSPAC data) was significantly stronger than the estimated confounded association due to shared genetics.

# Causal effect of fetal growth on gestational duration, maternal BP, and FPG

To test the possible fetal drive of fetal growth on birth outcomes and maternal pregnancy phenotypes, we constructed genetic scores using 86 SNPs associated with birth weight with confirmed fetal effect [27] and tested their associations with birth outcomes as well as maternal BP and FPG during pregnancy (Tables <u>3</u> and <u>4</u>). The fetal genetic score for birth weight was significantly associated with gestational-age–adjusted birth weight with an  $R^2 = 3.1\%$ . The paternal transmitted haplotype score ( $S_{h3}$ ) consistently has larger effect than the maternal transmitted score ( $S_{h1}$ ) (<u>S15 Table</u>), probably because of a negative maternal effect of the same alleles on



Estimated differences in adult phenotypes (SD)

**Fig 5. Comparing the magnitudes of genetically confounded associations (differences in adult phenotypes [in SDs] per 1-SD difference in birth weight) with observed and reported phenotypic associations.** Confounded (eff): genetically confounded associations estimated based on the fetal genetic effect on birth weight (adjusted by gestational duration) of the variants associated with an adult phenotype. Confounded (cor): genetically confounded associations between birth weight (unadjusted by gestational duration) and adult phenotype based on the reported genetic correlations [20]. Observed: observed phenotypic associations between gestational-duration–adjusted birth weight and offspring height, BMI, BP, and FPG measured at age 17 in ASLPAC. Reported: the reported associations between birth weight (unadjusted by gestational duration) and BP and T2D susceptibility from a recent epidemiological study [53]. One-SD values in the gestational-duration–adjusted and unadjusted birth weight were 426 g and 484 g, respectively. BMI, body mass index; BP, blood pressure; FPG, fasting plasma glucose; SD, standard deviation; T2D, type 2 diabetes.

https://doi.org/10.1371/journal.pmed.1003305.g005

birth weight and birth length, as shown by the negative effect of the nontransmitted haplotype score  $(S_{h2})$  (Table 3). The maternal and paternal transmitted birth weight haplotype scores  $(S_{h1})$  and  $S_{h3}$ ) were also associated with shorter gestational duration and increased preterm birth risk (Table 3 and S16 Fig). Using the paternal transmitted haplotype score as an instrument, the estimated causal effects were 3.2 days (95% CI: 1.5 to 5.0,  $p = 2.7 \times 10^{-4}$ ) shorter gestation and an approximate doubling of the preterm birth risk per 1-SD changes in fetal growth rate (Table 3). In addition, we also observed significant associations between paternal transmitted

Genetic Score Association and	Ges	tational	Days	Preterm	n Birth (lo	g[OR])	B	Birth W	eight (g)	Bi	rth Lengtl	h (cm)
Causal Estimation <sup>a</sup>	Beta	SE	p-Value	Beta	SE	<i>p</i> -Value	Beta	SE	p-Value	Beta	SE	<i>p</i> -Value
Maternal transmitted ( $\beta_{h1}$ )	-0.0056	0.0027	0.035*	0.0024	0.00064	$0.00018^{*}$	0.79	0.087	$1.80\times10^{-19_\ast}$	0.0018	0.00048	0.00022*
Maternal nontransmitted ( $\beta_{h2}$ )	0.001	0.0027	0.7	-0.00058	0.00065	0.37	-0.19	0.088	0.029*	-0.00022	0.00049	0.66
Paternal transmitted ( $\beta_{h3}$ )	-0.0099	0.0026	0.00018*	0.0019	0.00064	0.003*	1.3	0.087	$1.70\times10^{-48_*}$	0.0031	0.00048	$2.30\times10^{-10*}$
Maternal effect ( $\beta_{MY}$ )	0.0026	0.0023	0.26	$-5.80\times10^{-5}$	0.00055	0.92	-0.33	0.076	$1.10\times10^{-5*}$	-0.00073	0.00042	0.082
Fetal effect ( $\beta_{FY}$ )	-0.0083	0.0023	0.00029*	0.0025	0.00056	$9.20\times 10^{-6_{\ast}}$	1.1	0.075	$1.00\times10^{-50*}$	0.0025	0.00042	$1.30\times10^{-9_{\ast}}$
Causal (TSLS)	-2.92	0.805	0.00029*	0.837	0.267	0.0017*		N	A	1.05	0.124	$2.40\times10^{-17*}$
Causal (ratio)	-3.24	0.889	0.00027*	0.624	0.215	0.0036*				1	0.172	$5.10 \times 10^{-9_{*}}$

Table 3. Associations between birth weight genetic scores and birth outcomes and estimated causal effects per 1-SD change in gestational-age-adjusted birth weight.

<sup>a</sup>The effect size (beta) and SEs of genetic score association were given by per unit (g) change in genetic scores; the causal effect sizes were based on per 1-SD (1 SD = 426 g) change in gestational-age–adjusted birth weight. **Abbreviations:** NA, not applicable; SD, standard deviation; SE, standard error; TSLS, two-stage least-squares. \**p*-Values less than 0.05.

https://doi.org/10.1371/journal.pmed.1003305.t003

birth weight score ( $S_{h3}$ ) and maternal SBP ( $p = 2.2 \times 10^{-2}$ ) measured during pregnancy and the estimated causal effects (ratio method) were 1.4 mmHg (95% CI: 0.2 to 2.6,  $p = 2.3 \times 10^{-2}$ ) increase in SBP per 1-SD changes in fetal growth rate (<u>Table 4</u> and <u>S17 Fig</u>).

## Discussion

In this report, we utilized a haplotype-based genetic score approach to explicitly model the maternal and fetal genetic effects of maternal phenotypes and pregnancy outcomes (Fig 2). From the estimated maternal and fetal genetic associations (Table 2), we estimated maternal causal effects and the genetically confounded associations between maternal phenotypes and birth outcomes (Fig 3), as well as the genetically confounded associations between birth

Table 4. Associations between birth weight genetic scores and maternal BP, glucose levels, and estimated causal effects per 1-SD changes in gestational-age–adjusted birth weight.

Genetic Score Association	SBI	P <sup>b</sup> (mmH	Ig)	DBP	<sup>b</sup> (mmH	[g)	FPG	(mmol/L	.)
and Causal Estimation <sup>a</sup>	Beta	SE	<i>p-</i> Value	Beta	SE	<i>p-</i> Value	Beta	SE	<i>p-</i> Value
Maternal transmitted ( $\beta_{h1}$ )	-0.0026	0.0019	0.17	-0.00021	0.0013	0.87	-0.00011	0.00022	0.61
Maternal nontransmitted $(\beta_{h_2})$	0.00057	0.0019	0.77	0.0006	0.0013	0.65	-0.00025	0.00022	0.26
Paternal transmitted ( $\beta_{h3}$ )	0.0043	0.0019	0.022*	0.0021	0.0013	0.1	$8.90  imes 10^{-5}$	0.00022	0.69
Maternal effect ( $\beta_{MY}$ )	-0.0032	0.0016	0.053	-0.00083	0.0011	0.47	-0.00022	0.0002	0.26
Fetal effect ( $\beta_{FY}$ )	0.00054	0.0016	0.74	0.00065	0.0011	0.57	0.00011	0.00019	0.55
Causal (TSLS)	1.2	0.551	0.03*	0.641	0.378	0.09	0.0203	0.0565	0.72
Causal (ratio)	1.41	0.621	0.023*	0.69	0.426	0.1	0.029	0.0724	0.69

<sup>a</sup>The effect size (beta) and SEs of genetic score association were given by per unit (g) change in genetic scores; the causal effect sizes were based on per 1-SD (1 SD = 426 g) change in gestational-age-adjusted birth weight.

Abbreviations: ALSPAC, The Avon Longitudinal Study of Parents and Children; BP, blood pressure; DBP, diastolic blood pressure; FPG, fasting plasma glucose; HAPO, Hyperglycemia and Adverse Pregnancy Outcome; SBP, systolic blood pressure; SD, standard deviation; SE, standard error; TSLS, two-stage least-squares.

<sup>b</sup>Maternal BP (SBP and DBP) in ALSPAC and HAPO.

<sup>c</sup>FPG during pregnancy measured in HAPO only.

\**p*-Values less than 0.05.

https://doi.org/10.1371/journal.pmed.1003305.t004



**Fig 6. Estimated maternal and fetal genetic effects underlying the associations between maternal phenotypes and pregnancy outcomes and their associations with adult phenotypes.** Blue arrows: maternal or fetal genetic effects. Green arrows: maternal causal effects. Red arrows: "fetal drive." Dashed arrows: genetically confounded associations between birth outcomes and late adult phenotypes due to shared genetics. The widths of the arrows were approximately proportional to the estimated effect sizes evaluated by per-SD changes, which can be found in <u>S13 Table</u> (maternal causal effect), Tables <u>3</u> and <u>4</u> (fetal drive), and <u>Fig 4</u> (genetically confounded associations). BMI, body mass index; BP, blood pressure; FPG, fasting plasma glucose; SD, standard deviation; T2D, type 2 diabetes

https://doi.org/10.1371/journal.pmed.1003305.g006

outcomes and adult phenotypes (Fig 4). We also tested whether fetal growth (as indicated by gestational-age–adjusted birth weight) could influence gestational duration, maternal BP, and maternal FPG levels during pregnancy (Tables 3 and 4). Our results revealed complex maternal and fetal genetic effects in shaping the associations between maternal phenotypes and birth outcomes and their associations with adult phenotypes (Fig 6).

Our results support that maternal height influences the duration of gestation and fetal growth, and alleles associated with adult height also influence birth size through fetal genetic effects. These new results support our previous finding [15] with additional evidence. We observed evidence supporting causal effect of maternal BMI on birth weight and length. Similar observations have been reported before [24]. We found no evidence demonstrating that SNPs associated with BMI have significant fetal genetic effects on birth weight or length. We also observed that alleles elevating BP are associated with shorter gestational duration though a maternal effect but are associated with reduced fetal growth through a fetal genetic effect. The later finding is consistent with Horikoshi and colleagues [20]; however, it is inconsistent with Warrington and colleagues [27], who did not find a fetal effect of alleles for BP on fetal growth. We showed a positive maternal effect of FPG on fetal growth. This result is consistent with previous studies [5, 24]. However, the glucose-increasing alleles in the fetus are associated with reduced birth weight, which is in line with the epidemiological finding that paternal diabetes is associated with lower birth weight [54] and with the fetal insulin hypothesis [19]. The T2D-

associated SNPs can influence birth weight in a similar way as FPG SNPs but with a weaker maternal effect and a stronger fetal effect. These observations indicate that the maternal causal effect is mainly driven by maternal FPG levels. The fetal genetic effect is likely mediated by fetal insulin, as proposed by the fetal insulin hypothesis [19]. By utilizing the birth weight genetic score built on paternal transmitted alleles, our findings also support causal effects of fetal growth on gestational duration and maternal BP.

These findings have several implications. First, maternal size and fetal growth are important factors in defining the duration of gestation. This is demonstrated not only by the evidence supporting a causal effect of maternal height (size of the mother) on gestational duration, but also by the observation that the maternal or fetal genetic effects on fetal growth are usually associated with opposing effects on gestational duration. For example, rapid fetal growth due to either high maternal FPG or direct fetal genetic effects of growth-promoting alleles shortens gestational duration. Whether this "trade-off" between fetal growth and gestational duration is due to physical [55] or metabolic [56] constraints will require further investigation. Alleles associated with BP have negative impacts on both birth weight and gestational duration but mainly through either fetal genetics or a maternal effect, respectively.

Secondly, fetal growth (as evaluated by gestational-duration-adjusted birth weight and length) is influenced by both maternal and fetal effects. In addition to the positive maternal effect of maternal FPG on birth weight, the many alleles associated with body height, BP, FPG level, or T2D susceptibility can influence fetal growth through fetal genetic effects. The alleles associated with body height are positively associated with birth size, whereas the alleles associated with higher metabolic risks (for example, high BP, blood glucose level, and higher risk of T2D) reduce birth weight, which also suggests that lower birth weight (or, more precisely, small for gestational age status) might be a predictor of the load of genetic metabolic risks. Our results show that the shared genetic effects largely explain the life-course associations between birth weight and many cardiometabolic phenotypes (Fig 5), a result that is consistent with the reported inverse genetic correlations between birth weight and late-onset metabolic disorders [20], and both support a strong genetic rather than an environmental effect underlying the life-course association between birth weight and later metabolic risks. Compared with the previous analyses based on genetic correlations using genome-wide SNPs, our approach estimated the life-course associations by extrapolating the effects of the top GWA SNPs associated with the adult phenotypes and therefore has a specific mechanistic implication-the life-course associations were mainly driven by genetic variants with large effects on adult phenotypes rather than by the variants with large effects on fetal growth. In addition, our method can partition the genetically confounded associations to either maternal effects or shared genetic effects in offspring (S1 Text).

The results from this study also reveal a major theme in human pregnancy—both maternal effects and direct fetal genetic effects jointly shape the observed associations between maternal phenotypes and pregnancy outcomes. The same genetic variants can influence different birth outcomes or the same birth outcomes through both maternal and fetal effects, and these 2 types of effects can be antagonistic, as exemplified by the opposing maternal and fetal effects of the FPG associated alleles on birth weight, or the effects of height-associated alleles on gestational duration. These complex mechanisms can be further complicated by "fetal drive," as shown by the associations between paternal transmitted birth weight genetic score and gestational duration as well as maternal BP, i.e., fast fetal growth shortens gestational duration and increases maternal BP (Tables 3 and 4).

Our study had a number of limitations. First, there were some incomplete and heterogeneous phenotype data. Most maternal phenotypes (for example, age, height, and BMI) and birth outcomes (for example, gestational duration and birth weight) were available from the study data sets; however, FPG level during pregnancy was only available in HAPO, and BP data were only available from HAPO and ALSPAC. Another limitation is that because of incomplete data, we were not able to include important environmental or socioeconomic factors in the analysis. Some of the maternal exposures (for example, maternal smoking) are known to be associated with both maternal phenotypes and birth outcomes [57], which may introduce confounding. However, we argue that the genetic scores used in our study are not known to be associated with these factors, and therefore, our analyses are robust to potential confounding due to these environmental factors [58]. Biological pleiotropy is always an issue in causal inference using genetic instruments, especially when a large number of genetic variants are used [45, 59]. We used the MR-PRESSO method to detect and remove SNPs with horizontal pleiotropic effects (S1 Text). The results (S16 Table and S17 Table) were essentially the same as the corresponding haplotype genetic score associations (Tables 2, 3 and 4). There is some evidence of pleiotropy (MR-PRESSO global test p < 0.05, S18 Table and S19 Table), suggesting the identified maternal effects may be not exclusively mediated by the targeted maternal phenotypes and the fetal effects of the SNPs on a birth outcome are not always proportional to their report effects on the adult phenotype (S4 Fig). After excluding the outliers, the corrected estimates and the *p*-values did not change substantially, and all the MR-PRESSO distortion test *p*-values were nonsignificant (S18 Table, S19 Table, and S18 Fig).

Multiple maternal and fetal phenotypes were examined in this study. The heritability (S2 <u>Table</u>) and the variance explained by their corresponding genetic scores (S4 <u>Table</u>) varied across these phenotypic traits. Because of these differences in variance being explained depending upon phenotype, we had different powers in testing the causal effects or genetic correlations among these traits. In the analysis of causal effects of fetal growth on maternal BP and FPG during pregnancy (<u>Table 4</u>), the phenotype data were only available in a subset of samples, which compromises the power of statistical analyses. These results will benefit from further replication in independent cohorts. Genetic variants associated with birth weight were used as genetic instrument for fetal growth. However, birth weight is an endpoint of fetal growth, and it cannot capture the temporal changes and effects of fetal growth during the course of pregnancy. Nevertheless, because the major aim of the study is to distinguish and to compare the relative contributions of maternal and fetal effects, we believe we have provided robust evidence in support of our major conclusions (Fig 6).

To conclude, our study revealed that many SNPs associated with maternal height, BP, and blood glucose levels (or T2D susceptibility) can have various maternal and fetal genetic effects on gestational duration and fetal growth. These maternal and fetal genetic effects may explain the observed associations between the studied maternal phenotypes and birth outcomes, as well as the life-course associations between these birth outcomes and adult phenotypes. Our findings related to gestational-age–adjusted birth weight suggest that rapid fetal growth might reduce gestational duration and increase maternal BP. These findings provide additional insights into the mechanisms behind the observed associations between maternal phenotype and birth outcomes and their life-course impacts on later-life health. Although our current study focused on pregnancy outcomes measured at birth (for example, gestational duration and birth weight), similar analysis can be conducted on longitudinal measures (for example, those related to growth pattern in early life and later adulthood). With the accumulation of more longitudinal birth cohorts, the dissection of maternal and fetal genetic effects may open up future opportunities to explore how maternal effect, fetal development, and genetics influence long-term health and well-being.

### Supporting information

S1 STROBE Checklist. STROBE checklist. STROBE, Strengthening the Reporting of Obser-

vational Studies in Epidemiology. (PDF)

S1 Text. Supplementary methods.

(PDF)

**S1 Data. GWA SNPs used in constructing genetic scores.** GWA, genome-wide association; SNP, single-nucleotide polymorphism. (XLSX)

**S1 Table. Data sets and number of samples.** (PDF)

**S2 Table. GWA SNPs used to build the genetic scores.** GWA, genome-wide association; SNP, single-nucleotide polymorphism. (PDF)

**S3** Table. Association between maternal phenotypes and birth outcomes. (PDF)

**S4** Table. Associations between maternal genetic scores and maternal traits. (PDF)

S5 Table. Associations between maternal height genetic scores and maternal height. (PDF)

**S6 Table. Associations between maternal BMI genetic scores and maternal BMI.** BMI, body mass index. (PDF)

**S7 Table.** Associations between maternal BP genetic scores and maternal BP (SBP and DBP). BP, blood pressure; DBP, diastolic BP; SBP, systolic BP. (PDF)

**S8 Table.** Associations between maternal FPG and T2D genetic scores and maternal FPG. FPG, fasting plasma glucose; T2D, type 2 diabetes. (PDF)

S9 Table. Correlation (p-value) between maternal genotype score ( $S_{mat}$ ), fetal genotype score ( $S_{fet}$ ), and the 3 haplotype genetic scores:  $S_{h1}$ ,  $S_{h2}$ , and  $S_{h3}$ . (PDF)

S10 Table. Association between haplotype genetic scores and birth outcomes—meta-analyses using random-effects model. (PDF)

**S11 Table.** Association between haplotype genetic scores and birth outcomes based on the ALSPAC data set only. ALSPAC, The Avon Longitudinal Study of Parents and Children. (PDF)

**S12 Table.** Association between haplotype genetic scores and birth outcomes based on meta-analysis of the FIN, MoBa, DNBC, HAPO, and GPN data sets. DNBC, The Danish National Birth Cohort; FIN, The Finnish birth data set; GPN, The Genomic and Proteomic Network for Preterm Birth Research; HAPO, Hyperglycemia and Adverse Pregnancy

Outcome; MoBa, The Mother Child data set of Norway. (PDF)

**S13** Table. Phenotypic associations, estimated maternal causal effects, and genetically confounded associations (per 1-SD changes in maternal traits) between maternal traits and birth outcomes. SD, standard deviation. (PDF)

S14 Table. Estimated changes in adult phenotypes per 1-SD changes in birth outcomes. SD, standard deviation.

(PDF)

S15 Table. Association between fetal birth weight genetic score and gestational-ageadjusted birth weight (gram). (PDF)

**S16 Table. MR-PRESSO: Effects of maternal traits on birth outcomes.** MR-PRESSO, mendelian randomization pleiotropy residual sum and outlier. (PDF)

**S17 Table. MR-PRESSO: effects of fetal growth on birth outcomes (A) and maternal BP and glucose levels (B).** BP, blood pressure; MR-PRESSO, mendelian randomization pleiotropy residual sum and outlier. (PDF)

**S18 Table. MR-PRESSO global test, outlier test, and distortion test results: effects of maternal traits on birth outcomes.** MR-PRESSO, mendelian randomization pleiotropy residual sum and outlier. (PDF)

**S19 Table.** MR-PRESSO global test, outlier test, and distortion test results: effects of fetal growth on birth outcomes (A) and maternal BP and glucose levels (B). BP, blood pressure; MR-PRESSO, mendelian randomization pleiotropy residual sum and outlier.

(PDF)

**S1 Fig. Processing of phenotype and genotype data.** (PDF)

**S2** Fig. General representation of confounded association. (PDF)

S3 Fig. Genetically confounded association between a maternal phenotype and a birth outcome (A) and between a birth outcome and an adult phenotype in offspring (B) because of shared genetics.

(PDF)

**S4 Fig. Multivariable MR analysis of maternal and fetal genetic effect.** MR, mendelian randomization.

(PDF)

**S5 Fig. Distributions of maternal height, weight, and BMI.** BMI, body mass index. (PDF)

**S6 Fig. Distributions of maternal BP and glucose levels.** BP, blood pressure. (PDF)

**S7** Fig. Distributions of gestational days, birth weight, and birth length. (PDF)

**S8** Fig. Estimated effect sizes of adult height haplotype scores on pregnancy outcomes. (PDF)

**S9 Fig. Estimated effect sizes of BMI haplotype scores on pregnancy outcomes.** BMI, body mass index.

(PDF)

**S10 Fig. Estimated effect sizes of BP haplotype scores on pregnancy outcomes.** BP, blood pressure.

(PDF)

**S11** Fig. Estimated effect sizes of blood glucose haplotype scores on pregnancy outcomes. (PDF)

**S12 Fig. Estimated effect sizes of T2D haplotype scores on pregnancy outcomes.** T2D, type 2 diabetes.

(PDF)

**S13** Fig. Estimated sizes of causal effects and genetically confounded associations per 1-SD changes in maternal traits on preterm birth risk and birth length. SD, standard deviation. (PDF)

S14 Fig. Estimated differences in adult phenotypes (SD) per 1-SD difference in gestational days (left) and preterm birth risk (right). SD, standard deviation. (PDF)

S15 Fig. Estimated differences in adult phenotypes (SD) per 1-SD difference in gestational-age-adjusted birth weight (left) and birth length (right). SD, standard deviation. (PDF)

**S16 Fig. Estimated effect sizes of birth weight haplotype scores on pregnancy outcomes.** (PDF)

**S17 Fig. Estimated effect sizes of birth weight haplotype scores on maternal BP.** BP, blood pressure.

(PDF)

**S18 Fig. An example of multivariable MR estimate.** MR, mendelian randomization. (PDF)

#### Acknowledgments

We thank the participants in the Finnish birth cohort as well as the research group who collected the data. We are grateful to all the participating families in Norway who took part in the MoBa cohort study. We are extremely grateful to all the families who took part in ALSPAC, the midwives for their help in recruiting them, and the whole ALSPAC team, which includes interviewers, computer and laboratory technicians, clerical workers, research scientists, volunteers, managers, receptionists, and nurses. We are very grateful to all DNBC families who took part in the study. We would also like to thank everyone involved in data collection and biological material handling. We would like to acknowledge the participants and research personnel at the participating HAPO field centers. We thank the infants and their parents who agreed to take part in the GPN study and the medical and nursing colleagues who collected that data. We thank dbGAP for depositing and hosting the phenotype and genotype data of the DNBC, HAPO, and GPN data sets.

#### **Author Contributions**

Conceptualization: Louis J. Muglia, Ge Zhang.

Data curation: Jonas Bacelis, Amit Srivastava, Amy Rouse, Mikko Hallman, Ge Zhang.

- Formal analysis: Jing Chen, Ge Zhang.
- Funding acquisition: Louis J. Muglia, Ge Zhang.
- **Investigation:** Jing Chen, Rachel M. Freathy, Deborah A. Lawlor, Jeffrey C. Murray, Scott M. Williams, Bo Jacobsson, Louis J. Muglia, Ge Zhang.
- Methodology: Jing Chen, Jonas Bacelis, Bjarke Feenstra, Ge Zhang.
- Project administration: Bo Jacobsson, Louis J. Muglia, Ge Zhang.
- **Resources:** Jonas Bacelis, Mikko Hallman, Kari Teramo, Mads Melbye, Bjarke Feenstra, George Davey Smith, Deborah A. Lawlor, Jeffrey C. Murray, Bo Jacobsson, Louis J. Muglia, Ge Zhang.
- Software: Jing Chen, Ge Zhang.
- Supervision: Bo Jacobsson, Louis J. Muglia, Ge Zhang.
- Validation: Jonas Bacelis, Pol Sole-Navais, Julius Juodakis, Ge Zhang.
- Visualization: Jing Chen, Rachel M. Freathy, Jeffrey C. Murray, Ge Zhang.
- Writing original draft: Jing Chen, Rachel M. Freathy, Louis J. Muglia, Ge Zhang.
- Writing review & editing: Jing Chen, Jonas Bacelis, Pol Sole-Navais, Amit Srivastava, Julius Juodakis, Mikko Hallman, Kari Teramo, Bjarke Feenstra, Rachel M. Freathy, George Davey Smith, Deborah A. Lawlor, Jeffrey C. Murray, Scott M. Williams, Bo Jacobsson, Louis J. Muglia, Ge Zhang.

#### References

- Han Z, Lutsiv O, Mulla S, McDonald SD. Maternal height and the risk of preterm birth and low birth weight: a systematic review and meta-analyses. J Obstet Gynaecol Can. 2012; 34(8):721–46. <u>https:// doi.org/10.1016/S1701-2163(16)35337-3</u> PMID: <u>22947405</u>.
- Myklestad K, Vatten LJ, Magnussen EB, Salvesen KA, Romundstad PR. Do parental heights influence pregnancy length?: A population-based prospective study, HUNT 2. BMC Pregnancy Childbirth. 2013; 13:33. <u>https://doi.org/10.1186/1471-2393-13-33</u> PMID: <u>23383756</u>; PubMed Central PMCID: PMC3608172.
- Dougherty CR, Jones AD. The determinants of birth weight. Am J Obstet Gynecol. 1982; 144(2):190– 200. https://doi.org/10.1016/0002-9378(82)90627-5 PMID: 7114129.
- Addo OY, Stein AD, Fall CH, Gigante DP, Guntupalli AM, Horta BL, et al. Maternal height and child growth patterns. J Pediatr. 2013; 163(2):549–54. <u>https://doi.org/10.1016/j.jpeds.2013.02.002</u> PMID: 23477997; PubMed Central PMCID: PMC3711792.
- HAPO Study Cooperative Research Group, Metzger BE, Lowe LP, Dyer AR, Trimble ER, Chaovarindr U, et al. Hyperglycemia and adverse pregnancy outcomes. N Engl J Med. 2008; 358(19):1991–2002. https://doi.org/10.1056/NEJMoa0707943 PMID: 18463375.
- Steer PJ, Little MP, Kold-Jensen T, Chapple J, Elliott P. Maternal blood pressure in pregnancy, birth weight, and perinatal mortality in first births: prospective study. BMJ. 2004; 329(7478):1312. <u>https://doi.org/10.1136/bmj.38258.566262.7C</u> PMID: <u>15561733</u>; PubMed Central PMCID: PMC534837.
- Himmelmann A, Svensson A, Hansson L. Relation of maternal blood pressure during pregnancy to birth weight and blood pressure in children. The Hypertension in Pregnancy Offspring Study. J Intern Med. 1994; 235(4):347–52. <u>https://doi.org/10.1111/j.1365-2796.1994.tb01085.x</u> PMID: <u>8151267</u>.

- Parsons TJ, Power C, Logan S, Summerbell CD. Childhood predictors of adult obesity: a systematic review. Int J Obes Relat Metab Disord. 1999; 23 Suppl 8:S1–107. PMID: 10641588.
- Whincup PH, Kaye SJ, Owen CG, Huxley R, Cook DG, Anazawa S, et al. Birth weight and risk of type 2 diabetes: a systematic review. JAMA. 2008; 300(24):2886–97. <u>https://doi.org/10.1001/jama.2008.886</u> PMID: 19109117.
- Falkner B. Birth weight as a predictor of future hypertension. Am J Hypertens. 2002; 15(2 Pt 2):43S–5S. https://doi.org/10.1016/s0895-7061(01)02297-x PMID: 11866228.
- Huxley R, Owen CG, Whincup PH, Cook DG, Rich-Edwards J, Smith GD, et al. Is birth weight a risk factor for ischemic heart disease in later life? Am J Clin Nutr. 2007; 85(5):1244–50. <u>https://doi.org/10.1093/</u> ajcn/85.5.1244 PMID: 17490959.
- Miura K, Nakagawa H, Tabata M, Morikawa Y, Nishijo M, Kagamimori S. Birth weight, childhood growth, and cardiovascular disease risk factors in Japanese aged 20 years. Am J Epidemiol. 2001; 153(8):783– 9. https://doi.org/10.1093/aje/153.8.783 PMID: <u>11296151</u>.
- Lawlor D, Richmond R, Warrington N, McMahon G, Davey Smith G, Bowden J, et al. Using Mendelian randomization to determine causal effects of maternal pregnancy (intrauterine) exposures on offspring outcomes: Sources of bias and methods for assessing them. Wellcome Open Res. 2017; 2:11. <u>https:// doi.org/10.12688/wellcomeopenres.10567.1</u> PMID: <u>28405635</u>; PubMed Central PMCID: PMC5386135.
- Lawlor DA. The Society for Social Medicine John Pemberton Lecture 2011. Developmental overnutrition—an old hypothesis with new importance? Int J Epidemiol. 2013; 42(1):7–29. <u>https://doi.org/10.</u> 1093/ije/dys209 PMID: 23508404
- Zhang G, Bacelis J, Lengyel C, Teramo K, Hallman M, Helgeland O, et al. Assessing the Causal Relationship of Maternal Height on Birth Size and Gestational Age at Birth: A Mendelian Randomization Analysis. PLoS Med. 2015; 12(8):e1001865. <u>https://doi.org/10.1371/journal.pmed.1001865</u> PMID: 26284790; PubMed Central PMCID: PMC4540580.
- Smith GD. Assessing intrauterine influences on offspring health outcomes: can epidemiological studies yield robust findings? Basic Clin Pharmacol Toxicol. 2008; 102(2):245–56. <u>https://doi.org/10.1111/j.</u> <u>1742-7843.2007.00191.x</u> PMID: <u>18226080</u>.
- Barker DJ. The origins of the developmental origins theory. J Intern Med. 2007; 261(5):412–7. <u>https://</u> doi.org/10.1111/j.1365-2796.2007.01809.x PMID: <u>17444880</u>.
- Martin-Gronert MS, Ozanne SE. Mechanisms underlying the developmental origins of disease. Rev Endocr Metab Disord. 2012; 13(2):85–92. <u>https://doi.org/10.1007/s11154-012-9210-z</u> PMID: 22430227.
- Hattersley AT, Tooke JE. The fetal insulin hypothesis: an alternative explanation of the association of low birthweight with diabetes and vascular disease. Lancet. 1999; 353(9166):1789–92. <u>https://doi.org/ 10.1016/S0140-6736(98)07546-1</u> PMID: 10348008.
- Horikoshi M, Beaumont RN, Day FR, Warrington NM, Kooijman MN, Fernandez-Tajes J, et al. Genome-wide associations for birth weight and correlations with adult disease. Nature. 2016; 538 (7624):248–52. https://doi.org/10.1038/nature19806 PMID: 27680694.
- Wolf JB, Wade MJ. What are maternal effects (and what are they not)? Philos Trans R Soc Lond B Biol Sci. 2009; 364(1520):1107–15. <u>https://doi.org/10.1098/rstb.2008.0238</u> PMID: <u>19324615</u>; PubMed Central PMCID: PMC2666680.
- Liu N, Archer E, Srinivasasainagendra V, Allison DB. A statistical framework for testing the causal effects of fetal drive. Front Genet. 2014; 5:464. <u>https://doi.org/10.3389/fgene.2014.00464</u> PMID: 25628644; PubMed Central PMCID: PMC4292723.
- Power C, Kuh D, Morton S. From developmental origins of adult disease to life course research on adult disease and aging: insights from birth cohort studies. Annu Rev Public Health. 2013; 34:7–28. <u>https:// doi.org/10.1146/annurev-publhealth-031912-114423</u> PMID: 23514315.
- Tyrrell J, Richmond RC, Palmer TM, Feenstra B, Rangarajan J, Metrustry S, et al. Genetic Evidence for Causal Relationships Between Maternal Obesity-Related Traits and Birth Weight. JAMA. 2016; 315 (11):1129–40. <u>https://doi.org/10.1001/jama.2016.1975</u> PMID: <u>26978208</u>; PubMed Central PMCID: PMC4811305.
- Smith GD, Ebrahim S. Mendelian randomization: prospects, potentials, and limitations. Int J Epidemiol. 2004; 33(1):30–42. <u>https://doi.org/10.1093/ije/dyh132</u> PMID: <u>15075143</u>.
- Richmond RC, Timpson NJ, Felix JF, Palmer T, Gaillard R, McMahon G, et al. Using Genetic Variation to Explore the Causal Effect of Maternal Pregnancy Adiposity on Future Offspring Adiposity: A Mendelian Randomisation Study. PLoS Med. 2017; 14(1): e1002221. <u>https://doi.org/10.1371/journal.pmed.</u> <u>1002221</u> PMID: <u>28118352</u>; PubMed Central PMCID: PMC5261553.

- Warrington NM, Beaumont RN, Horikoshi M, Day FR, Helgeland O, Laurin C, et al. Maternal and fetal genetic effects on birth weight and their relevance to cardio-metabolic risk factors. Nat Genet. 2019; 51 (5):804–14. <u>https://doi.org/10.1038/s41588-019-0403-1</u> PMID: <u>31043758</u>; PubMed Central PMCID: PMC6522365.
- Bacelis J, Juodakis J, Adams Waldorf KM, Sengpiel V, Muglia LJ, Zhang G, et al. Uterine distention as a factor in birth timing: retrospective nationwide cohort study in Sweden. BMJ Open. 2018; 8(10): e022929. <u>https://doi.org/10.1136/bmjopen-2018-022929</u> PMID: <u>30385442</u>.
- Kramer MS. The epidemiology of adverse pregnancy outcomes: an overview. J Nutr. 2003; 133(5 Suppl 2):1592S–6S. <u>https://doi.org/10.1093/jn/133.5.1592S.</u> PMID: <u>12730473</u>.
- Zhang G, Srivastava A, Bacelis J, Juodakis J, Jacobsson B, Muglia LJ. Genetic studies of gestational duration and preterm birth. Best Pract Res Clin Obstet Gynaecol. 2018; 52:33–47. <u>https://doi.org/10. 1016/j.bpobgyn.2018.05.003</u> PMID: <u>30007778</u>; PubMed Central PMCID: PMC6290110.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gotzsche PC, Vandenbroucke JP, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. PLoS Med. 2007; 4(10):e296. <u>https://doi.org/10.1371/journal.pmed.0040296</u> PMID: 17941714; PubMed Central PMCID: PMC2020495.
- Zhang G, Feenstra B, Bacelis J, Liu X, Muglia LM, Juodakis J, et al. Genetic Associations with Gestational Duration and Spontaneous Preterm Birth. N Engl J Med. 2017; 377(12):1156–67. <u>https://doi.org/ 10.1056/NEJMoa1612665</u> PMID: <u>28877031</u>; PubMed Central PMCID: PMC5561422.
- Fraser A, Macdonald-Wallis C, Tilling K, Boyd A, Golding J, Davey Smith G, et al. Cohort Profile: the Avon Longitudinal Study of Parents and Children: ALSPAC mothers cohort. Int J Epidemiol. 2013; 42 (1):97–110. Epub 2012 Apr 16. <u>https://doi.org/10.1093/ije/dys066</u> PMID: <u>22507742</u>; PubMed Central PMCID: PMC3600619.
- Zhang H, Baldwin DA, Bukowski RK, Parry S, Xu Y, Song C, et al. A genome-wide association study of early spontaneous preterm delivery. Genet Epidemiol. 2015; 39(3):217–26. <u>https://doi.org/10.1002/ gepi.21887</u> PMID: 25599974; PubMed Central PMCID: PMC4366311.
- Macdonald-Wallis C, Tilling K, Fraser A, Nelson SM, Lawlor DA. Established preeclampsia risk factors are related to patterns of blood pressure change in normal term pregnancy: findings from the Avon Longitudinal Study of Parents and Children. J Hypertens. 2011; 29(9):1703–11. <u>https://doi.org/10.1097/</u> <u>HJH.0b013e328349eec6</u> PMID: <u>21841545</u>.
- Das S, Forer L, Schonherr S, Sidore C, Locke AE, Kwong A, et al. Next-generation genotype imputation service and methods. Nat Genet. 2016; 48(10):1284–7. <u>https://doi.org/10.1038/ng.3656</u> PMID: 27571263; PubMed Central PMCID: PMC5157836.
- 1000 Genomes Project Consortium, Abecasis GR, Auton A Brooks LD, Durbin RM Garrison EP, Kang HM, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. <u>https://doi.org/10.1038/nature15393</u> PMID: 26432245.
- Delaneau O, Marchini J, Zagury JF. A linear complexity phasing method for thousands of genomes. Nat Methods. 2012; 9(2):179–81. https://doi.org/10.1038/nmeth.1785 PMID: 22138821.
- Yengo L, Sidorenko J, Kemper KE, Zheng Z, Wood AR, Weedon MN, et al. Meta-analysis of genomewide association studies for height and body mass index in approximately 700000 individuals of European ancestry. Hum Mol Genet. 2018; 27(20):3641–9. <u>https://doi.org/10.1093/hmg/ddy271</u> PMID: 30124842.
- Evangelou E, Warren HR, Mosen-Ansorena D, Mifsud B, Pazoki R, Gao H, et al. Genetic analysis of over 1 million people identifies 535 new loci associated with blood pressure traits. Nat Genet. 2018; 50 (10):1412–25. <u>https://doi.org/10.1038/s41588-018-0205-x</u> PMID: <u>30224653</u>; PubMed Central PMCID: PMC6284793.
- Manning AK, Hivert MF, Scott RA, Grimsby JL, Bouatia-Naji N, Chen H, et al. A genome-wide approach accounting for body mass index identifies genetic variants influencing fasting glycemic traits and insulin resistance. Nat Genet. 2012; 44(6):659–69. <u>https://doi.org/10.1038/ng.2274</u> PMID: <u>22581228</u>; PubMed Central PMCID: PMC3613127.
- Mahajan A, Taliun D, Thurner M, Robertson NR, Torres JM, Rayner NW, et al. Fine-mapping type 2 diabetes loci to single-variant resolution using high-density imputation and islet-specific epigenome maps. Nat Genet. 2018; 50(11):1505–13. <u>https://doi.org/10.1038/s41588-018-0241-6</u> PMID: <u>30297969</u>; PubMed Central PMCID: PMC6287706.
- Lawlor DA, Harbord RM, Sterne JA, Timpson N, Davey Smith G. Mendelian randomization: using genes as instruments for making causal inferences in epidemiology. Stat Med. 2008; 27(8):1133–63. <u>https://doi.org/10.1002/sim.3034</u> PMID: <u>17886233</u>.
- 44. Sivakumaran S, Agakov F, Theodoratou E, Prendergast JG, Zgaga L, Manolio T, et al. Abundant pleiot-ropy in human complex diseases and traits. Am J Hum Genet. 2011; 89(5):607–18. <u>https://doi.org/10.1016/j.ajhg.2011.10.004</u> PMID: <u>22077970</u>; PubMed Central PMCID: PMC3213397.

- 45. Evans DM, Brion MJ, Paternoster L, Kemp JP, McMahon G, Munafo M, et al. Mining the human phenome using allelic scores that index biological intermediates. PLoS Genet. 2013; 9(10): e1003919. https://doi.org/10.1371/journal.pgen.1003919 PMID: 24204319; PubMed Central PMCID: PMC3814299.
- 46. Burgess S, Scott RA, Timpson NJ, Davey Smith G, Thompson SG, Consortium E-I. Using published data in Mendelian randomization: a blueprint for efficient identification of causal risk factors. Eur J Epidemiol. 2015; 30(7):543–52. <u>https://doi.org/10.1007/s10654-015-0011-z</u> PMID: <u>25773750</u>; PubMed Central PMCID: PMC4516908.
- Lawlor DA. Commentary: Two-sample Mendelian randomization: opportunities and challenges. Int J Epidemiol. 2016; 45(3):908–15. <u>https://doi.org/10.1093/ije/dyw127</u> PMID: <u>27427429</u>; PubMed Central PMCID: PMC5005949.
- Verbanck M, Chen CY, Neale B, Do R. Detection of widespread horizontal pleiotropy in causal relationships inferred from Mendelian randomization between complex traits and diseases. Nat Genet. 2018; 50(5):693–8. <u>https://doi.org/10.1038/s41588-018-0099-7</u> PMID: <u>29686387</u>; PubMed Central PMCID: PMC6083837.
- Tyler AL, Asselbergs FW, Williams SM, Moore JH. Shadows of complexity: what biological networks reveal about epistasis and pleiotropy. Bioessays. 2009; 31(2):220–7. <u>https://doi.org/10.1002/bies.</u> 200800022 PMID: 19204994; PubMed Central PMCID: PMC3159922.
- Viechtbauer W. Conducting Meta-Analyses in R with the metafor Package. J Stat Softw. 2010; 36 (3):48. https://doi.org/10.18637/jss.v036.i03
- Robinson MR, Kleinman A, Graff M, Vinkhuyzen AAE, Couper D, Miller MB, et al. Genetic evidence of assortative mating in humans. Nature Human Behaviour. 2017; 1:0016. <u>https://doi.org/10.1038/ s41562-016-0016</u>
- Li X, Redline S, Zhang X, Williams S, Zhu X. Height associated variants demonstrate assortative mating in human populations. Sci Rep. 2017; 7(1):15689. <u>https://doi.org/10.1038/s41598-017-15864-x</u> PMID: 29146993; PubMed Central PMCID: PMC5691191.
- Knop MR, Geng TT, Gorny AW, Ding R, Li C, Ley SH, et al. Birth Weight and Risk of Type 2 Diabetes Mellitus, Cardiovascular Disease, and Hypertension in Adults: A Meta-Analysis of 7 646 267 Participants From 135 Studies. J Am Heart Assoc. 2018; 7(23): e008870. <u>https://doi.org/10.1161/JAHA.118.</u> 008870 PMID: <u>30486715</u>.
- 54. Tyrrell JS, Yaghootkar H, Freathy RM, Hattersley AT, Frayling TM. Parental diabetes and birthweight in 236 030 individuals in the UK biobank study. Int J Epidemiol. 2013; 42(6):1714–23. <u>https://doi.org/10.1093/ije/dyt220</u> PMID: <u>24336895</u>; PubMed Central PMCID: PMC3887570.
- 55. Washburn SL. Tools and human evolution. Sci Am. 1960; 203:63–75. PMID: 13843002.
- 56. Dunsworth HM, Warrener AG, Deacon T, Ellison PT, Pontzer H. Metabolic hypothesis for human altriciality. Proc Natl Acad Sci U S A. 2012; 109(38):15212–6. <u>https://doi.org/10.1073/pnas.1205282109</u> PMID: <u>22932870</u>; PubMed Central PMCID: PMC3458333.
- 57. Agrawal A, Scherrer JF, Grant JD, Sartor CE, Pergadia ML, Duncan AE, et al. The effects of maternal smoking during pregnancy on offspring outcomes. Prev Med. 2010; 50(1–2):13–8. Epub 2009 Dec 21. <a href="https://doi.org/10.1016/j.ypmed.2009.12.009">https://doi.org/10.1016/j.ypmed.2009.12.009</a> PMID: 20026103; PubMed Central PMCID: PMC2813884.
- Smith GD, Ebrahim S. 'Mendelian randomization': can genetic epidemiology contribute to understanding environmental determinants of disease? Int J Epidemiol. 2003; 32(1):1–22. <u>https://doi.org/10.1093/ ije/dyg070</u> PMID: <u>12689998</u>.
- Solovieff N, Cotsapas C, Lee PH, Purcell SM, Smoller JW. Pleiotropy in complex traits: challenges and strategies. Nat Rev Genet. 2013; 14(7):483–95. <u>https://doi.org/10.1038/nrg3461</u> PMID: <u>23752797</u>; PubMed Central PMCID: PMC4104202.